

Received 13 July 2022, accepted 27 July 2022, date of publication 1 August 2022, date of current version 10 August 2022. *Digital Object Identifier* 10.1109/ACCESS.2022.3195875

TOPICAL REVIEW

Next Generation Cognition-Aware Hearing Aid Devices With Microwave Sensors: Opportunities and Challenges

USMAN ANWAR^{®1}, TUGHRUL ARSLAN^{®1}, AMIR HUSSAIN², (Senior Member, IEEE), AND PETER LOMAX¹

¹School of Engineering, The University of Edinburgh, Edinburgh EH9 3FF, U.K.
²School of Computing, Edinburgh Napier University, Edinburgh EH10 5DT, U.K.

Corresponding author: Usman Anwar (usman.anwar@ed.ac.uk)

This work was supported in part by U.K. Engineering and Physical Sciences Research Council (EPSRC) under Grant EP/T021063/1; and in part by the Higher Education Commission (HEC), Pakistan.

ABSTRACT The strong association between hearing loss and cognitive decline has developed into a major health challenge that calls for early detection, diagnosis and prevention. Hearing loss usually results in severe health implications that include loss of mobility, communication problems and cognitive decline. This study provides an overview of the effects of hearing loss on cognition and progressive neurological disorders with a discussion on the future scope of microwave portable technologies in care homes arrangement. Moreover, the efficacy of hearing aids in reversing cognitive decline and dementia has been investigated. The interconnection between hearing loss, cognitive load and neurodegeneration is also explored. Furthermore, this study looks into the prospects of using portable microwave sensors for the detection and monitoring of cognitive load. For early detection of dementia, this study proposes the integration of microwave sensors with hearing aid devices. Implications and design challenges of portable antenna systems for neurodegeneration detection have also been considered. Future improvement areas regarding robust analysis and diagnosis, system accuracy and security, user-centricity and device privacy for a broader clinical implementation are also discussed.

INDEX TERMS Age-related hearing loss, cognitive load, dementia, mild cognitive impairment, microwave sensors, neurodegeneration, non-invasive, portable.

I. INTRODUCTION

More than 5% of the world's population suffers from hearing loss which is around 360 million people across the globe [1]. Hearing loss affects roughly 11 million people in the UK, making it the second most prevalent disability [2]. Hearing loss increases sharply with age, and affects more than 40% of adults over the age of 50 in the UK, growing to more than 70% over the age of 70 [3]. Around 75% of the old population in care homes is disproportionately affected by some form of hearing loss. Healthy ageing is linked to neurological and microvascular changes, which often means the start of

The associate editor coordinating the review of this manuscript and approving it for publication was Wen-Sheng Zhao^(b).

Age-Related Hearing Loss (ARHL) and cognitive decline at the same time [4]. It is important to address hearing loss immediately after diagnosis, to diminish the adverse impacts. There are several measures available for the rehabilitation of people with hearing loss which include the use of hearing aids, middle ear and cochlear implants. Hearing aids could benefit 6.7 million of the 8 million individuals in the UK, but only two million use them [5]. Unassisted hearing loss usually leads to severe health implications in older people which include social isolation, mobility loss and cognitive decline. In the case of unassisted hearing loss, people with minor hearing loss are twice as likely as those without hearing loss to suffer from cognitive decline. People with moderate hearing loss are three times more likely to suffer from dementia, while those with severe hearing loss are five times more likely.

Around six million people in the UK suffer from brain-related diseases each year. Major elements that lead to neurodegeneration include cognitive loading, dementia, Alzheimer's disease and stroke. People with dementia experience memory loss due to neurodegeneration affecting brain areas responsible for creating and retrieving memories. The gradual nature of dementia, starting with the mild earlystage symptoms of cognitive degeneration and the low diagnosis rate, makes it difficult to have an early diagnosis. Mild Cognitive Impairment (MCI) is the period between the cognitive decline of normal ageing and the serious decline of dementia [6]. MCI may increase the risk of dementia triggered by Alzheimer's disease or other neurodegenerative conditions [7]. MCI is triggered due to various underlying medical conditions that may include high blood pressure, diabetes and heart issues. These underlying conditions may result in abnormal blood flow to the brain, which in turn can affect the cerebral metabolism and result in an increase in cognitive load [8]. Delay in diagnosis usually results in progressive loss of brain neurons that causes cognitive decline in elderly patients. Early diagnosis of cognitive load and removal of underlying risks is important to slow down the risk of cognitive decline and dementia [9]. Some existing radiology technologies like Magnetic Resonance Imaging (MRI), Positron Emission Tomography (PET) and Computerized Tomography (CT) scans can detect this degeneration at advanced stages but require extensive medical supervision and are expensive. Radiofrequency and microwave sensors can be an effective substitute for these conventional medical technologies. Considerable research in biomedical applications for wearable sensors is done during the past few years. These are directed more towards Alzheimer's disease [10], [11], brain tumour [12], hemorrhagic stroke [13] and cancer [14] detection. There is not much work done on wearable healthcare applications for cognitive load detection and dementia. There is a research gap in terms of microwave portable sensors for cognitive load leading to dementia detection. These portable devices can be linked to hearing aid devices for the estimation of cognitive load inside the brain. This can help to avoid the progression of neurodegeneration at a preliminary stage. Early detection of cognitive decline will help to ease the burden on both medical and care home facilities. It is also important to investigate the link between hearing loss, cognitive load, and dementia. This is the prime motivation behind writing this literature review.

The rest of this paper is structured as follows: Section II discusses the causes, implications and challenges that arise from hearing loss. Section III gives an overview of hearing loss implications on neurological function and progressive diseases, with a focus on cognitive load and dementia. The interrelation between hearing loss, cognitive load and dementia is also discussed. The prospects of early cognitive load detection in reversing cognitive decline and the onset of dementia are investigated in Section IV. The role of hearing

aids in reversing cognitive decline is discussed in Section V. Section VI explores the factors that lead to high cognitive load and symptoms associated with it. Section VII provides a brief overview of major neurodegenerative diseases along with the latest statistics. Section VIII explains the scope of wearable microwave sensor systems in the detection of neurodegeneration, with a focus on dementia, stroke and Alzheimer's disease. Challenges and limitations of wearable microwave sensor systems concerning robust diagnosis, system security, accuracy, user-centricity and device privacy for a broader clinical practice are discussed in Section IX. Future research directions along with considerations for improvement are discussed in Section X. The review concludes by discussing the future areas and potential of smart microwave portable technologies in hospitals and care homes arrangement.

II. EFFECTS OF HEARING LOSS ON BRAIN

Hearing loss or impairment usually occurs among the elderly, and affects Activities of Daily Living (ADLs) [15]. Hearing loss causes a shift in cognitive resources from memory to auditory processing, putting an undue strain on brain functions [16]. In severe situations, this leads to cognitive decline and dementia.



FIGURE 1. Association hierarchy of ARHL with dementia.

Hearing loss usually coincides with other conditions including social isolation, anxiety, falls, depression, decreased quality of life, and memory and communication problems. ARHL has been linked to a quicker rate of cognitive decline [17]. There are multiple theories associated with it, first one is related to cognitive load that happens as a result of untreated hearing loss. Hearing loss is more common in adults with dementia [15]. According to recent studies, dementia and Alzheimer's disease are more likely to occur in older adults with hearing loss [17], [18]. The brain gets overworked by constant strain to understand audio and speech. More concentration is directed towards information deciphering and the brain becomes less focused on retaining that information. The second theory relates to social isolation and loneliness, untreated hearing loss causes a reduction in socializing [19]. Social isolation usually results in less stimulation of the brain. Brain cells, particularly those that hear and process sound, can decrease due to a lack of stimulation. This affects motor skills and leads to structural changes which cause the brain to shrink. Long-term auditory deprivation can affect cognitive performance by lowering communication quality, which can lead to social isolation, depression, and dementia [18]. Hearing loss is recurrently linked to cognitive deterioration in most studies. However, extensive research on the relationship between hearing impairment and MCI has only been undertaken in the last decade. Hearing impairment is linked to a higher risk of dementia and cognitive impairment [20].

Auditory deprivation results in social isolation that leads to depression and a decline in cognitive function. Long-term hearing loss shifts cognitive resources away from memory and toward auditory processing [16], putting an undue burden on higher cortical functions that eventually leads to cognitive decline. The complete hierarchy of ARHL with dementia is given in Fig.1. Hearing loss can alter the auditory pathway and affects the auditory brain, resulting in dementia and cognitive decline [16]. The other hypothesis is that neurodegenerative diseases attack the auditory brain, resulting in peripheral and central hearing loss [21]. These neurodegenerative pathologies include canonical dementias such as Lewy-body dementia, vascular dementia, Alzheimer's disease and frontotemporal dementia. Temporal, frontal, subcortical and parietal circuits that underpin auditory cognition are all compromised by these neurodegenerative pathologies [22]. The principal neurodegenerative causes of dementia in middle to later life are pathogenic protein distribution over large-scale cerebral networks and various forms of localized brain shrinkage. These are the key symptoms which can help to anticipate the hearing impairments which accompany these specific dementia conditions.

III. COGNITIVE DECLINE AND HEARING LOSS

Hearing loss causes degraded auditory signals, and auditory perceptual processing necessitates more cognitive resources/ This results in a cognitive shift away from other tasks and toward effortful listening [23]. This phenomenon finally leads to cognitive reserve depletion. The excessive cognitive load causes neurodegeneration with mild to severe brain changes [24]. Hearing impairment may be aggravated by this cognitive reserve depletion and this leads to degeneration in auditory perception [25]. Therefore, the attention required to understand and comprehend speech is vital for individuals with hearing loss.

Hearing loss, according to the cognitive load hypothesis, results in degraded auditory signals. Increased cognitive resources required for auditory perceptual processing, and diversion from other cognitive tasks to effortful listening, finally lead to cognitive reserve depletion [26], as shown in Fig. 2. According to this proposition, the ageing brain suffers from a neurodegenerative process that causes both cognitive impairment and hearing loss. Another study established that hearing loss leads to cognitive deterioration which is either permanent or reversible with rehabilitation [18]. Hearing loss increases cognitive strain in patients with cognitive impairment, according to this study. Impaired perception may lead to deterioration in cognition and social seclusion, both of which can contribute to cognitive decline. When a person has difficulty remembering things, learning new things, concentrating, or making decisions that influence their daily lives, they are said to be suffering from cognitive decline [27]. Cognitive decline levels usually increase from mild to severe. Changes in cognitive functions start after a mild cognitive decline, but it does not affect the everyday activities of the person suffering from it. Severe cognitive decline can lead to a loss of ability to talk, write, and comprehend things which result in dependence on others for daily activities, thus creating an inability to live independently [28].

Cognitive load is the amount of load being faced by the human cognitive system while going through a specific task. It represents the strain on cognitive resources such as working memory, brain processing, and visual and verbal information processing units. In cognitive psychology, the amount of working memory resources being utilized is referred to as cognitive load [29].

Several underlying factors which can be responsible for an increase in cognitive load include hearing loss, hypertension, hyperlipidemia, and cardiovascular and cerebrovascular disease [30], as illustrated in Fig. 3. Persistent high blood pressure levels result in arterial muscle hyperplasia and arterial stiffness which eventually leads to vascular remodelling and inflammation in cerebral blood vessels [31].

Other factors that contribute to the increase in cognitive load are chronic hypertension and hyperlipidemia. Hypertension affects small cerebral arteries which can lead to stroke and brain haemorrhage. Hypertension disturbs the mechanisms that regulate cerebral blood flow, in addition to changing the anatomy of cerebral blood vessels [32]. These alterations make it harder for brain metabolites like betaamyloid and tau-tangles to be cleared, rather than allow them to build up. These anatomical and functional abnormalities caused by hypertension increase cognitive load. This culminates in cerebrovascular dysfunction, making the brain more susceptible to degenerative pathologies like dementia and Alzheimer's disease [33]. Cognitive decline or MCI is a harbinger of severe neurological conditions such as Alzheimer's disease, dementia and stroke [34]. MCI is an intermediate stage between the possible cognitive decline of normal ageing and the more severe decline of dementia. MCI can increase the vulnerability to dementia caused by Alzheimer's disease or other neurodegenerative conditions [35]. Capturing the early signs of cognitive decline may help to delay the acute dementia development at later stages.

IV. DEMENTIA AND HEARING LOSS

Around 50 million people worldwide suffer from dementia, a figure that is expected to rise to 152 million by 2050 [36].



FIGURE 2. Effects of hearing loss on cognitive function.



FIGURE 3. Progression from hearing loss to cognitive impairment.

Furthermore, hearing loss affects over 465 million individuals, including one-third of those over the age of 65. According to the World Health Organization, untreated hearing loss costs the world economy \$750 billion per year, while the current annual cost of dementia is around \$1 trillion, with estimates that this could double by 2030 [37].

Hearing loss in middle age is supposed to contribute to 9% of dementia occurrences, and the latest statistics suggest that dementia affects 47 million people worldwide [38]. ARHL is usually caused by cochlear damage, while dementia occurs as a result of cortical degeneration with an initial deterioration in the multimodal cortex [39]. There are multiple studies which relate peripheral auditory decline to major cortical changes associated with dementia [24].

Recent studies have linked severe dementia to intensive hearing loss. This means that hearing loss has a linear relationship with dementia and that the risk of dementia increases multiple times with incremental severity of hearing loss [22]. Cognitive processing gradually slows down with age [40], and at the age of 70, one-fifth of the population is estimated to have a considerable degree of cognitive loss [41].

According to this study, ARHL impacts severely on the brain and is assumed to be a primary reason for cognitive decline in older adults. ARHL has a negative influence on cognitive performance and raises the risk of dementia by adding more damage to existing brain impairments such as tau-tangles [17], amyloid-beta, brain atrophies [42] and microvascular infarct [24]. Vascular brain anomalies, macro and microvascular infarction, aggravated as a result of ARHL, usually lead to vascular dementia and Alzheimer's in most cases. Some studies have suggested a strong correlation between hearing loss and stroke [43], [44]. Central white matter pathways degradation and lower cortical volumes in the primary auditory cortex are also linked to peripheral

hearing loss [45]. Cochlear losses may result in cortical reorganization and degeneration of stria vascularis [46].

Hearing impairment may result in rapid brain shrinkage and decreased primary auditory cortex volumes in elderly adults, and ARHL is commonly associated with reduced brain volume [47]. Grey matter volume reduction in the auditory cortex is suggested to be a probable cause of peripheral hearing loss in the elderly [42], [48]. Hearing ability and grey matter volume had a significant linear relationship [49], and individual differences in hearing capacity predicted the amount of neuronal activation in temporal gyri, comprising of auditory cortex, brainstem, and thalamus.

Central auditory processing is the ability of the brain to understand sounds received by the cochlea [3]. As a result, it is vulnerable to neurodegeneration, and data suggests that central auditory processing is impaired early in Alzheimer's disease and MCI [50]. Central auditory impairment is measured by dichotic listening activities and can be a precursor to Alzheimer's disease [51].

V. DO HEARING AIDS HELP IN REVERSING COGNITIVE DECLINE?

Hearing aids or a cochlear implant can help with social and emotional functioning, communication, and cognitive performance, as well as improve the overall quality of life [52]. Detecting the severity of hearing impairment is critical to slow down the disease progression. Routine hearing care can be provided to the general public to protect cognitive function and reduce the public health costs associated with MCI and dementia [53]. Hearing examinations can be used clinically to assess the risk of MCI/dementia, and hearing aids can be used to delay dementia in older people with hearing impairment [54].

The available treatments include hearing aids, middle ear and cochlear implants that can delay the onset of cognitive decline. Recent studies have confirmed that hearing aids are beneficial in the early stages of ARHL, and hearing aids can be helpful in the rehabilitation of higher cortical functions [55], [56]. The use of hearing aids helps to reverse central auditory system ageing [4]. Statistical survey analysis proved that hearing aids can be an effective tool to slow down the conversion from MCI to dementia [57]. But the role of hearing aids in vacating high-level processing resources and improving cognitive function is not fully explored. Considerable improvements are observed in cognition with hearing aid usage and the results returning to baseline where the hearing aid is not involved in human subjects.

VI. COGNITIVE LOAD, DIAGNOSIS AND REPERCUSSIONS

Listening effort to comprehend speech results in higher utilization of cognitive resources for hearing impaired individuals. The depletion of cognitive reserve causes an undue cognitive strain on neurological functions [58].

There are various other factors contributing to cognitive load. Major causes include age-related hearing loss, cardiovascular disease, depression, and hypertension [59]. These underlying causes eventually lead to disruption of blood flow to the brain and affect cerebral metabolism. Cognitive load, if left untreated, may progress to dementia in later stages [60].

Preliminary cognitive load symptoms can be associated with a change in blood glucose pattern for the brain, which is directly linked to changes in blood flow pattern [61]. This could appear before any noticeable neuron loss and reduction in grey matter volume. Changes in blood perfusion levels, as well as the target location within the brain, can help in the differentiation between different kinds of dementia [62]. Recent studies have discovered a positive linear association between cognitive load and blood pressure [12]. An increase in blood pressure can lead to a high cognitive load [13]. As a result, different blood pressure levels can be used to assess the amount of cognitive load. Increased systolic blood pressure is easy to detect and is more likely to have an impact on the human body. Because blood pressure and metabolic rate are related in a positive linear way [63], the metabolic rate is also a major indicator to gauge the cognitive loading state.

VII. NEURODEGENERATION: ETIOLOGIES AND CHALLENGES

Neurons are the basic building blocks of the human nervous system and constitute a major part of the brain. Neurons do not reproduce or regenerate themselves, and cannot be replaced by the body in case of damage due to neurodegenerative diseases [64]. Neurodegenerative diseases are often incurable and result in progressive degeneration, which cannot be reversed but can be slowed down if diagnosed at an early stage [65]. Major degenerative diseases include Dementia, Parkinson's, Huntington's, Alzheimer's and motor neuron disease. Ageing is a common risk factor in all of these, and increasing life expectancy worldwide is resulting in the sharp growth of neurodegenerative diseases [66]. Factors like agerelated hearing loss, hypertension, and microvascular and cardiovascular anomalies increase the likelihood of cognitive decline and neurodegeneration.

Alzheimer's disease is the most common type of neurodegeneration and the sixth largest cause of death in the United States, with 5.5 million people aged 65 and older living with Alzheimer's disease [67]. Around 50 million people worldwide suffer from dementia which is estimated to increase to 152 million by 2050 [68]. The distribution of major neurodegenerative diseases according to recent statistics is provided in Fig. 4. The annual cost of dementia is almost \$1 trillion, according to the World Health Organization (WHO), with estimates that this could double by 2030 [69]. An increase in the ageing population is leading to a considerable financial and socioeconomic burden on both individuals and caregivers.

Beta-amyloid and tau-tangles are two toxic proteins that cause cellular brain damage in Alzheimer's disease [73]. Beta-amyloid plaques form between neurons and disrupt the neurotransmitter receptors, making it difficult for the cell to operate and relay messages to neighbouring neurons [74]. Beta-amyloid also interferes with other proteins



FIGURE 4. Distribution of major neurodegenerative diseases, based on statistics from [36], [67], [70], [71], [72].

in the hippocampus that is a crucial component of the brain for memory [75]. As the neuron's function is reduced, fewer neurotransmitters are produced, and communication between neurons decreases. Tau tangles are twisted protein strands found inside neurons that cause cell death by preventing cells from receiving essential nutrients [76]. The damage induced by beta-amyloid plaques and tau tangles eventually kills brain cells. The effects of these plaques and tangles are most noticeable during the middle stage of dementia.

Vascular dementia is the second most common form of dementia after Alzheimer's disease and is also one of the leading causes of death today [77]. Around 17 percent of people above 65 years of age with dementia have vascular dementia [78]. Vascular dementia typically starts with a decreased blood flow to the brain due to clots or vascular blockage in various segments of the brain [79]. This leads to vascular impairment, infarcts, white matter lesions and brain atrophy in some cases. Initial pathological changes occur in the white matter of the brain which is followed by micro and macro-infarcts [80]. In some cases, this leads to brain atrophy due to cerebrovascular impairment and long-term reduced blood flow in the brain. The damage caused by the infarction is mostly irreversible, however, a timely diagnosis can help avoid future cerebrovascular incidents.

VIII. PORTABLE MICROWAVE SENSORS FOR COGNITIVE LOAD AND NEURODEGENERATION DETECTION

In today's health sector, expensive medical imaging applications like MRI, ultrasound, PET, and CT scan tests are extensively deployed. Medical practitioners often use these procedures to obtain images of internal organs to diagnose disorders. These traditional technologies require more time and high exposure to the radiations to generate the imaging report. Imaging results produced by these invasive methods vary in terms of operating method, radiation exposure time, resolution, and implementation cost, as shown in Table 1. Even though these procedures are widely deployed, they are not flexible and depend on ionizing radiations which may have adverse effects on exposure for a longer duration [81]. cost, flexible, compact, non-invasive, low-exposure and nonionizing solution. Microwave sensing technology holds the potential to replace existing imaging equipment in the near future. Electromagnetic (EM) waves in the microwave frequency region are non-ionizing and can penetrate tissue, making them ideal for diagnostic applications [82]. Several techniques have been reported recently for the detection of tumours, brain stroke, Alzheimer's and breast cancer [11], [12], [14], [83]. However, high temporal resolution is required for functional neuroimaging to detect abrupt variations in cerebral hemodynamics [82]. Microwave sensing techniques operate based on differences in dielectric permittivity and electrical conductivity between unhealthy and healthy regions of the human body. Electromagnetic (EM) waves are transmitted towards the intended area and reflected signals are received back to the sensors for diagnosis and analysis of any potential disease in that area of the body. The received signals are scattered and reflected with varying intensity depending on the difference in electrical properties of that body area. In the case of multiple antenna sensors, an antenna array can be used to send radiofrequency pulses to the target area of the human brain. Microwave signals are reflected through backscattering, received by the array and then analyzed using relevant software to detect if there is an active degeneration. The antenna sensors should be able to transmit signals over a wide range of frequencies to obtain high resolution and accurate images in such microwave imaging systems using the Ultra-wideband (UWB). Recent research on the detection of neurodegenera-

tion through microwave sensors is summarized in Table 2.

Microwave sensing and imaging systems have gained much attention in recent years as they provide a low-

A. COGNITIVE LOAD

Cognitive load detection is crucial to avoid the progression of neurodegenerations like stroke, carotid artery disease, Parkinson's disease and dementia. Various parameters can be utilized to assess cognitive load states. These include heart rate, blood pressure, skin conductance [84], [85], pupillometry [86] and thermal imaging [87]. However, the most accurate monitoring of the cognitive load can be achieved from the brain using sensors like Positron Emission Tomography (PET), Electroencephalogram (EEG) and head-mounted devices. PET scan produces detailed three-dimensional images to determine blood flow, oxygen levels, and metabolic variations. PET scanner detects the radiations returned by injected radiotracer inside the body [88]. It provides a detailed overview of metabolic changes appearing at the cellular level in tissue or an organ and is thus considered superior to CT scans and MRI [89]. EEG techniques use electrodes on the scalp to record the electrical response of the brain. Ionic motions in and around neurons generate these electrical signals during the activation and deactivation of neurons involved in cognitive tasks. The varying voltages in these electrical signals are measured by the EEG [90].

TABLE 1. Comparison of conventional medical imaging technologies.

Imaging Technologies	Computed Tomography (CT)		
Features	Scan	Magnetic Resonance Imaging (MRI)	Ultrasound
Principle of operation	Multiple X-rays from various angles to generate 3D cross- sectional imaging	Powerful magnetic fields and radiofrequency pulses are utilized for imaging	High-frequency sound waves to examine soft tissues such as muscles and internal organs
Applications and benefits	Detection of solid tumours, Superior bone tissue contrast compared to MRI	Detection of brain abnormalities and diagnosing soft-tissue injuries, More detailed imaging compared to CT Scan and Ultrasound	Real-time 2D imaging, Fast and sedation is not required for patients
Operation and maintenance cost	Less expensive than MRI	More expensive than CT scan and Ultrasound	Cheaper than CT scan and MRI
Scan time	5 to 10 minutes, depending on the area under scan	15 minutes to 2 hours, depending on the area being examined	15 to 45 minutes, depending on the area under scan
Radiation exposure	High radiation, Chest CT is equivalent to 100 chest X-rays	No ionizing radiation	No ionizing radiation
Effects on the human body	Ionizing radiation, Sedation or anaesthesia is required in some cases	Sedation or anesthesia is required in some cases, Requires patient to remain still for half-hour or more	Sedation or anesthesia is not required
Degree of invasiveness	Low	Low (But noisy and may cause claustrophobia, due to the enclosed space of the imaging machine)	Low
Disadvantages	Harmful for unborn babies Potential reaction to the use of dyes	More expensive than a CT scan, Slow image acquisition Expensive device	Lower image quality than CT scan, with effectiveness highly dependent on technician skills, Poor tissue contrast

Near-Infrared Spectroscopy (NIRS) and MRI can be utilized for cerebral blood flow measurement but both technologies are expensive and require extensive medical supervision. Functional Magnetic Resonance Imaging (fMRI) measures brain activity and neurovascular coupling through the detection of blood flow variations [91]. Similarly, the dielectric contrast between grey matter and blood can be utilized for microwave functional neuroimaging. Limited work is available in the literature for the detection of cognitive loading through microwave sensors. A UWB impulse radar is presented in [82] to detect blood volume in the cerebral cortex. Reflection from phantom liquid was detected and the attenuation was determined across a range of frequencies in the UWB band. Other than the head-mounted devices, chest-mounted and wrist-worn devices are also presented in the literature.

Several solutions have been presented recently for cognitive load monitoring using physiological parameters like galvanic skin response, heart rate, skin temperature and heart rate variation. Physiological sensors for measuring these indicators are low-cost, non-intrusive and readily available. Various studies have presented a linear relationship between galvanic skin response and cognitive load as the cognitive load increases with an increase in skin response [84]. Cardiac measurements including heart rate and heart rate variations are also utilized in some studies for estimation of cognitive load. Cognitive stress is measured remotely using a digital camera [92] and physiological parameters like breathing rate, heart rate, and heart rate variation are correlated with the sensors data to obtain an accuracy of more than 85%. Cognitive load was determined in driving conditions using heart rate variability [93]. It was established that the combination of skin conductance and heart rate measurements can result in a more accurate prediction of cognitive load.

Eye-tracking technology is another non-intrusive way of cognitive load estimation through monitoring of eye features and movement. Different estimation metrics presented in the literature are based on eye fixations, involuntary eye movements, frequent blinking and eye pupil oscillations. Longer eye fixations have been associated with a high cognitive load state in a few studies [94]. However, some studies indicated the opposite by correlating longer eye fixation, rapid eye movement is linked to high visual load and thus associated with higher cognitive load [96]. A high cognitive load state is also linked to more frequent eye blinking [97]. Cognitive load is also considered to have a persistent influence on eye pupil oscillations and pupil diameter [98]. Neuroimaging techniques can provide accurate detection of cognitive load but the technology is relatively expensive and not readily available to patients. Detection of cognitive load through neuro-imaging and Electroencephalography (EEG) is investigated in recent studies [99], [100]. Nearinfrared spectroscopy (NIRS) and EEG were combined to estimate the reaction and information processing time in intense cognitive load conditions [101]. The results were recorded through analysis of hemodynamics from the prefrontal, parietal and occipital lobe areas of the brain.

B. DEMENTIA

Despite improvement in lifestyle and reductions in risk factors such as cardiovascular anomalies, hypertension and hyperlipidemia, the cerebrovascular disease remains a threat, especially among the elderly. Factors such as protein deposits and reduced blood flow can result in degeneration of cortex and hippocampus atrophy [102], [103]. This could lead to mild or severe stage Dementia. This results in shrinkage of the hippocampus, enlargement of the brain's fluid-filled spaces and reduction in brain size. Dementia is an umbrella term to represent several diseases that affect memory, cognition, judgment and communication. Fig. 5 represents the dementia disease variations along with the major symptoms that lead to dementia.

The brain with dementia cannot receive its normal amount of blood and oxygen [104]. This reduces blood flow to the brain and damage brain cells [105]. Numerous underlying factors lead to dementia. These include high blood pressure, hyperlipidemia, diabetes, cardiovascular disease and stroke. Degeneration starts in the white matter of the brain which leads to brain atrophy and infarcts. The degeneration from infarction is irrevocable, but the progression of the disease can be restricted with an early diagnosis.

Timely diagnosis and detection of dementia are critical to avoid brain damage. Brain imaging is an essential tool in the diagnosis and determines the future treatment options available to the patient. Currently, CT-Scan, MRI and carotid ultrasound are being used for brain imaging and vascular dementia diagnosis [106], [107]. But these technologies are expensive, require extensive medical supervision and are not easily accessible, which causes substantial delays in diagnosis and results in irreversible damage.

Radio Frequency and microwave technologies can provide rapid sensing and diagnosis through compact, low-cost, noninvasive and wearable sensors. Major physiological changes can be captured at the preliminary stage using the reflection coefficient and dielectric variations. These wearable microwave sensors can detect the grey and white matter lesions, micro bleed, and micro and macro infarcts which leads to brain atrophy in the case of vascular, Lewy-body and frontotemporal dementias.

Dementia diagnosis requires cognitive assessment along with brain neuroimaging. Brain neuroimaging can detect structural degeneration including focal atrophy, tumour, bleeding and infarcts [108]. PET scans and functional neuroimaging are the most common methods for diagnosis of structural degeneration in vascular, Lewy-body and frontotemporal dementia. Due to the invasive and complex nature of these traditional scanning technologies, microwave sensing can be an inexpensive alternative.

An eight-element square monopole antenna with metasurface superstrate is presented in [109]. A blood mimicking target is identified in hemorrhagic stroke conditions and the system was designed to operate in the 0.5 GHz to 2 GHz band. A flexible, wideband, low-profile 8-element antenna array is proposed in [83] with an embedded Electromagnetic Band Gap (EBG) and metamaterial unit cells reflector. Antennas were designed on a multilayer low-cost, transparent, low loss and robust polymer Poly-Di-Methyl-Siloxane (PDMS) substrate that was optimized to operate in contact with the human head. The system comprised of a 4×4 radiating patch and a 10-unit EBG cell array around the feeding network that effectively suppressed the surface waves and improved the antenna impedance bandwidth. This antenna system was experimentally verified and tested on a three-dimensional (3D) head phantom with real human head properties. Accurate detection of infarction and bleeding was made using a confocal imaging algorithm.

A wearable device with integrated flexible microwave antennas was proposed to detect the progression of brain atrophy and lateral ventricle enlargement [11]. The device had an operating frequency range of 800 MHz to 2.5 GHz and was designed with a thin flexible Polyethylene Terephthalate (PET) substrate. Brain atrophy and lateral ventricle enlargement were detected and correlated using reflection and transmission coefficients. The device was tested on a realistic human head model with skin, skull, blood, and grey and white matter. Brain atrophy was simulated through a uniform reduction in the size of white and grey matter. The gaps were filled with cerebrospinal fluid. The wearable device was able to detect different levels of brain atrophy and lateral ventricle enlargement in both reflection and transmission modes. Similarly, changes in microwave reflection patterns can be utilized to diagnose lesions, atrophies and infarcts inside the frontal/temporal, white/grey matter of the brain, using wearable microwave devices.

C. ALZHEIMER'S DISEASE

Alzheimer's disease is the most common neurodegenerative disease and is also the sixth leading cause of death in the United States [110]. Alzheimer's disease is the most common cause of dementia in the United Kingdom. In the UK, Alzheimer's disease affects around 591,480 people out of estimated 954,000 dementia patients [111]. These figures are expected to increase to 1.5 million by 2040. Alzheimer's disease is a progressive neurodegenerative disorder that causes brain shrinkage, and atrophies and eventually results in neuron death [112].

MRI and PET scans are the primary monitoring techniques in use today, for Alzheimer's disease progression. Cerebrospinal fluid (CSF) testing is also performed in some cases to obtain evidence of Alzheimer's disease [108]. There is not much work done on wearable healthcare applications for cognitive load detection and dementia. However, considerable research exists in wearable healthcare applications for brain stroke and Alzheimer's detection. Six ultra-wideband monopole bidirectional antenna elements arranged symmetrically inside a wearable hat is presented in [11] to detect brain atrophy. Changes in brain volume and cerebrospinal fluid were captured using variation in reflection coefficient measurements. The design was simulated and experimentally verified using a real lamb brain inside a human head phantom.

Similarly, a six-element stepped monopole wideband antenna was designed to detect beta-amyloid plaques and tau tangles in the brain, for the diagnosis of Alzheimer's disease [10]. In addition to Alzheimer's disease detection, the dielectric properties of human brain tissue were investigated and an imaging algorithm was presented to reconstruct images of tangles and plaques. Relative permittivity and conductivity were measured for brain tissues affected by Alzheimer's disease and the results were compared with healthy brain tissues [113]. Tissues with a considerable amount of beta-amyloid plaques and tau tangles were collected from both grey matter and white matter of the frontal cortex. Results indicated an increase in conductivity and decrease in dielectric permittivity due to the presence of plaques and tangles, compared to the results from healthy brain tissues. Conductivity increased significantly with an increase in frequency from 100 MHz to 3 GHz, for both grey matter and white matter. The outcomes of this study can be beneficial in the detection of other neuropathologies and neurodegeneration.

D. BRAIN STROKE

Stroke is the fourth leading cause of death in the UK with more than 100,000 cases every year [120]. Age is an important factor for stroke and it is most likely to occur after the age of 55, but younger people including children also seem to be affected according to the latest statistics [121], [122]. A stroke occurs as a result of disruption in blood flow to the brain and can result in lifelong disabilities and death in extreme cases. In addition to the timely diagnosis and treatment, risk factors need to be managed for the prevention of stroke. High blood cholesterol and lipids result in plaque buildup inside arteries and can contribute to the thickening of arteries. This decreases the amount of blood flow to the brain, which can lead to Ischemic stroke [123]. Chronic high blood pressure damages the blood vessels and arteries which supply blood to the brain. Blood spills out in the brain due to damage to blood vessels and results in Hemorrhagic stroke [124].

A stroke is a medical emergency and an early diagnosis is crucial for prompt treatment [125]. Brain imaging is essential to localize the bleeding and clots. Traditional imaging technologies like CT Scan, MRI, Computed Tomographic Angiography (CTA), Magnetic Resonance Angiography (MRA) and carotid ultrasound can detect the damage to brain cells, bleeding and blood flow inside the arteries [126].



FIGURE 5. Dementia variations and major symptoms.

Although these technologies are extensively recommended for an accurate diagnosis of stroke, the invasiveness, high cost, low availability, long scan time, and expensive operation and maintenance make these procedures less viable for stroke patients.

In recent years, microwave sensing and imaging applications have been proposed and developed for sensing and imaging brain stroke and tumours. A Vivaldi antenna for stroke detection is proposed in [127], which was designed to operate between 1 GHz and 4 GHz. Multiple elements array of Vivaldi antennas were designed and interconnected using



TABLE 2. Comparison of microwave-based imaging solutions for neurodegeneration.

Sr. No.	Case Study	Health Area	RF and Microwave Technology	Operating Frequency	Key Findings
1	Razzicchia et al, 2021 [109]	Hemorrhagic stroke detection	8-element Printed square monopole antenna, enhanced by a metasurface superstrate	Operable in 0.5-2 GHz band	Possibility to detect target without using a bulky immersion liquid. MTS superstrate loading enhances the reflection coefficient and signals difference due to the presence of the target. A blood mimicking target is identified, and results are improved by MTS superstrate loading.
2	Alqadami et al, 2019 [83]	Hemorrhagic stroke detection	4 x 4 antenna array with symmetric U-slots, enhanced by mushroom-like 10- unit EBG around the feeding network to reduce surface waves	The operational frequency range of 1-2 GHz	Imaging results validated the capability of using the designed array system for the detection of bleeding in the brain using a confocal image algorithm. The designed system is flexible, low-profile, wideband, and unidirectional with Electromagnetic Band Gap (EBG) and metamaterial (MTM) unit cells reflector.
3	Bashri et al, 2018 [114]	Hemorrhagic stroke, brain injury	An antenna array of 12 wideband monopole antennas are used for the detection of brain injury and blood clots inside the brain. The reflection and transmission coefficient of antennas are collected for healthy and unhealthy brains. Use of low-power microcontroller with RF switches to control them via Bluetooth connection	The operational frequency range of 0.5-4 GHz	Blood clots are successfully detected by applying the confocal delay-and-sum imaging technique. Low cost, compact and lightweight RF switching system for wearable head imaging applications. Non-invasive and remote monitoring and imaging with this wearable device is possible using Bluetooth connectivity.
4	Bashri et al, 2017 [115]	Hemorrhagic stroke detection	8-element wearable flexible antenna array for stroke detection	The operational frequency range of 1.3-3.5 GHz	Capable of detecting stroke with flexible and lightweight antenna structure. Good reflection coefficient results over the designed ultra- wideband frequency range. Mutual coupling between adjacent antenna elements is less than -20 dB and SAR is within an acceptable range, 1.7 W/kg per 10g of tissue mass.
5	Alqadami et al, 2020 [13]	Brain stroke	16-element antenna array integrated with a flexible room temperature vulcanizing silicone cap	The operational frequency range of 0.6-2.5 GHz	Capability to detect and reconstruct 2D images of bleeding inside the brain. Enhanced antenna-skin matching with a flexible matching layer inside the cap. Challenges of size, rigidity and complex structures of existing systems are addressed.
6	Alqadami et al, 2021 [116]	Brain stroke	24-element planar antenna array, configured in two elliptical rings. Dielectric properties and favourable mechanical flexibility are enhanced by a flexible electromagnetic cap of customized polymer-ceramic composite Flexible matching layer on antenna array apertures, for enhanced impedance matching with skin	The operational frequency range of 0.9-2.5 GHz	Detection capability is experimentally verified on 3D head phantom with different stroke and imaging scenarios. 3D and 2D images using the beamforming and polar sensitivity encoding (PSE) image processing algorithms verifies the system's suitability for onsite brain imaging
7	Amor Smida, 2020 [117]	Brain stroke	Five different antenna designs are presented for brain stroke detection. SAR modelling and bioheat transfer analysis of the proposed system	The operational frequency range of 2.5 GHz	Simulated results confirmed the feasibility of stroke detection with the proposed antenna system and present high viability for portable, low-cost, and rapid stroke detection applications.
8	Saied et al, 2019 [11]	Alzheimer's disease	Six monopole bidirectional antenna elements are arranged symmetrically inside a wearable hat. Simulated and experimentally verified to detect brain atrophy, using a real lamb brain inside the human head phantom	Specific Absorption Rate between 0.0115-0.135 W/Kg and Operable in 0.8 to 3 GHz band	Capable of detecting various levels of brain atrophy and lateral ventricle enlargement. Results are verified by replacement of the brain's outer layer with Cerebral Spinal Fluid (CSF) and insertion of a cavity in the sample model
9	Saied et al, 2020 [10]	Alzheimer's disease	Six ultra wideband antennas are designed and optimized around the head phantom. Dielectric properties are analyzed first for the grey and white matter, to recreate various stages of Alzheimer's disease including Mild Cognitive Impairment	Dielectric measurements were obtained from grey and white matter regions of brain tissues with severe Alzheimer's disease pathology, within the frequency of 200 MHz to 3 GHz	Capable of Non-invasive monitoring and a novel wearable tool for medical diagnostic devices. Simulations and experimental results confirm that the designed system can detect different levels of brain atrophy and lateral ventricle enlargement in both the reflection and transmission modes. A promising tool for monitoring patients with Alzheimer's disease
10	Rezacieh et al, 2015 [12]	Brain tumour Detection	3D slot-rotated antenna for microwave head-imaging stem for detection of brain tumour. Parasitic patches are connected to slot area for enhancement of operational bandwidth	The operational frequency range of 1.4-3.5 GHz	Able to detect tumour target, results are verified on tumour target with 1 cm radius inside the artificial phantom. Compact size, wide operational bandwidth, unidirectional radiation.
11	Vasquez et al, 2020 [118]	Brain stroke	24-element helmet-shaped antenna array is presented for brain stroke imaging, with low complexity monopole antennas enclosed in graphite-silicon material	The operational frequency of 1GHz	Detected stroke target placed inside the liquid phantom, the system is validated through its digital twin which shows simulated data consistency with measured data. Full 3D brain imaging for detection of stroke target.
12	Sohani et al, 2020 [119]	Haemorrhagic stroke detection	Two antennas are embedded in an appropriate cavity, antennas are designed to rotate around the head and measure transmission and reflection data for capturing the stroke target.	The operational frequency range of 1-2 GHz	A haemorrhagic stroke target is detected using an anthropomorphic human head model, operable at very low input power. A portable device that is rotatable around the head for complete imaging of the brain. For imaging, the rotation subtraction artefact removal method is used in simulation and measurements.

microwave coaxial switches. This Vivaldi antenna array system could only be used for immobile diagnosis and was not portable. A flexible eight-element monopole antenna array is presented in [115], operable at frequencies between 1.3 GHz and 3.5 GHz. Antennas were arranged in an elliptical configuration on a thin flexible Polyethylene Terephthalate (PET) substrate, which makes them suitable for wearable applications. A flexible electromagnetic cap with an integrated 16-element antenna array is presented in [13]. It was designed with an operational frequency range of 0.6 to 2.5 GHz. This design was compact, light-weight and antenna elements were placed in a flexible multilayer wearable cap. An imaging algorithm was developed to process the collected data, and detection capability was verified using stroke-like targets in a realistic head phantom.

A 3D electromagnetic head imaging system with 24 elements planar antenna array is presented in [116] for stroke diagnosis. The designed array was arranged in two separate elliptical rings inside a compact flexible cap. A flexible layer was added to the front apertures of the antenna array to enhance impedance matching with the skin. Results were experimentally verified on realistic 3D head phantom using the beamforming and polar sensitivity encoding imaging algorithms. A portable 16-element antenna array is proposed in [128] for head imaging and stroke diagnosis. The antennas were designed on a 4mm room-temperature-vulcanizing (RTV) silicon substrate, operable on 0.45 to 3.6 GHz frequency band. Due to the meander-line structure, antenna elements were compact as compared to [13] and [116]. The generated images from testing on an artificial head phantom confirmed the viability of the system for stroke detection and diagnosis.

IX. PRACTICAL IMPLICATIONS OF PORTABLE MICROWAVE SENSORS

In contrast to traditional technologies, microwave systems are considered safe for the human body and can be employed in medical applications. Microwave sensors can provide noninvasive, low-radiation imaging that employs the external scattering field to measure the target. The most critical factor in microwave detection is precision. The antenna sensor system is responsible for transmitting and receiving signals in microwave imaging systems, hence its performance has a direct impact on the imaging effect. It is critical to create an antenna that is miniaturized for microwave imaging systems and may include features such as ultra-wide bandwidth, high gain, and good directivity. This requires design optimization to ensure that the antenna receives accurate data, keeping into consideration the suitability of design material for biomedical applications. Some of these design aspects are discussed below:

A. EASE OF USE AND USER-CENTRICITY

Portable gadgets have recently gained much attention due to their wide range of applications in sports, navigation, military, medical and space industry. A major bottleneck in the optimization of electronic components for portable applications is antenna design. These antennas must either be built into the human body, like wearables or incorporated within the clothing. They must be low-cost, low-profile, portable, resilient, reliable and low-power, or else they will be ineffective for these applications. The portable sensors should be immune to de-tuning and performance degradation caused by surrounding components when mounted on the body. The shadowing, scattering, fading and mismatching effects resulting from the human body and surrounding environment must be considered for wearable antennas [129].

Designing antennas for wearable applications can be challenging as the radiation efficiency is degraded considerably due to the higher dielectric constant of the body. There are various factors which need to be considered for user-centric design: the non-uniform structure of the human body, different blood concentrations and the composition of biological matter for each individual. The antenna performs differently for each human body depending upon the structure of internal organs, body mass, fat and muscle index [130]. For an efficient user-centric wearable device, some modelling constraints need to be considered. These include biological composition, body structure and varying electrical parameters of body matter. Another important factor that makes these portable devices more user-centric and conformable is the flexibility of antennas. The choice of non-rigid antennas for portable applications depends on the type of materials, substrate, processing technique and target application. The demand for flexible printed antennas has increased in recent years, specifically for biomedical applications [131], [132]. Flexible antennas are already being implemented in monitoring applications for neural interfaces, gait analysis, organ functions, vital signs and drug delivery systems [133]-[135]. The choice of substrate material is based on the dielectric properties and resistance to mechanical deformation like twisting, wrapping and bending of antennas [136].

B. DEVICE ACCURACY AND PRIVACY

Designing antenna sensors for portable applications is challenging as it requires careful consideration of antenna size, structure complexity and material of the dielectric substrate. Using free space antennas in radar-based systems can result in a severe mismatch between high permittivity skin and air with relatively lower permittivity [128]. To overcome this issue, additional air-skin medium and complicated calibration techniques are required to increase the matching between antenna elements and skin. Higher matching improves the penetration and detection capability for the localization of targets, particularly in the central part of the head. Effective matching requires a smooth transition of electromagnetic waves, for near-body applications. To compensate for the losses due to the high dielectric permittivity of the human body, successive layers of materials with increasing permittivity can be utilized for progressive adaption of electromagnetic waves between the air-tissue mediums. Two flat dielectric graded-index lenses were designed to increase the

matching between free space and tissues, for external hyperthermia [137]. This enhanced matching resulted in an effective propagation of electromagnetic waves within lossy and nonhomogeneous media.

Other parameters that need to consider for the design of microwave antenna sensors include reflection coefficient, peak gain, efficiency, Total Active Reflection Coefficient (TARC), Specific Absorption Rate (SAR), radiation pattern, impedance bandwidth and antenna polarization diversity.

Privacy and data safety risk mitigation in the sensing technology is of prime importance, as the imaging data gathered for analysis is vulnerable to privacy risks. Hence, the safety of patient data is mandatory and can be ensured by keeping the information in safe data storage systems. However, clear guidelines will be required for the safe operation of wearable devices and mitigation of data privacy concerns in domestic and care home setup.

C. ROBUST DIAGNOSIS

A low scan period and timely imaging results are critical for a robust diagnosis of neurodegeneration. Microwave sensors for biomedical applications are designed to use low frequencies for higher penetration. UWB antenna systems provide high capacity, high penetration and multipath robustness. Robust diagnosis is ensured by fine time resolution for accurate delay estimate. UWB antenna systems are inexpensive and utilize low transmission power with multi-access. For robust diagnosis through on-body microwave sensors, factors such as dispersion from human tissues across the frequency band need to be considered. Antennas with non-dispersive characteristics are required for optimal performance with radiation. This minimizes the path losses incurred through the UWB band.

D. ANTENNA MINIATURIZATION AT LOW FREQUENCIES

Designing antennas for portable applications is challenging as it involves antenna miniaturization to make it compatible with the wearable device. Antenna miniaturization helps to reduce the physical dimensions of the system while keeping the antenna functionality intact. For the miniaturization of antennas targeted at the elderly population, several design factors are required to be incorporated. These include ease-ofuse, non-intrusive, adaptability and portability. But there are various challenges involved in the miniaturization of antennas operating at lower frequencies. The antenna can be made smaller by increasing the frequency and decreasing the wavelength. But the wearable body sensors applications targeted at the brain require antenna sensors that operate at lower frequencies. Portable antennas for wearable applications must be robust against the bending and twisting effects in real-time applications [138].

E. MUTUAL COUPLING ISSUE BETWEEN ANTENNA ELEMENTS

To effectively scan the target areas of the body, the portable sensor system should contain multiple elements for extended coverage. Designing an antenna array on a thick and lossy substrate is critical, as it results in high mutual coupling between the elements. Mutual coupling can be dominant in two ways for these systems, either through surface waves or via free-space waves. Any of the unwanted couplings can dominate the other while designing an antenna array system and is dependent on the substrate width, ground plane, antenna type and the number of modes excited by the system. In printed patch antennas, the surface waves can be more dominant if the substrate thickness is according to this condition [139];

Substrate thickness
$$\geq \frac{0.3\lambda o}{2\pi \sqrt{\varepsilon_r}}$$
 (1)

where λo is the free space wavelength and ε_r is the relative permittivity of the substrate. Surface waves can result in a mutual coupling that causes degradation in antenna radiations. Surface wave propagation can be suppressed by modification in patch design, substrate and ground plane structure. Change in substrate material composition alters the relative permeability and permittivity which leads to variation in radiation parameters of the antenna [140].

Placement of antenna elements close to each other results in mutual coupling between them, which affects the overall gain and efficiency. Several techniques can be implemented to keep the spacing to an optimum level, these include the introduction of Defected Ground Structure (DGS), Electromagnetic Band Gap (EBG) [141], parasitic elements and the use of metamaterials to enhance isolation between elements, as shown in Fig. 6. Effects of the human body on portable Multiple Input Multiple Output (MIMO) antenna parameters like Envelope Correlation Coefficient (ECC), Mean Effective Gain (MEG) and channel capacity are to be considered in the design. For MIMO antenna systems, the required isolation between antenna elements is required to be more than 15 dB, the Envelope Correlation Coefficient (ECC) to be less than 0.05 and diversity gain to be less than 10 dB [138].

1) DEFECTED GROUND STRUCTURE (DGS)

DGS is implemented to miniaturize the design with an enhancement of bandwidth and gain. The fabrication process is relatively easier compared to other gain and bandwidth enhancement techniques. DGS structure can be realized by the addition of slots or defects on the ground plane. These etched defects in the ground plane change the effective inductance and capacitance of the microstrip line by adding slot resistance, inductance and capacitance [142]. Depending upon the shape, size and dimensions of the defect, the current distribution is affected in the ground plane which excites controlled propagation of EM waves through the substrate layer. DGS can be adopted to address several parameters of MIMO systems which includes low gain, cross-polarization and narrow bandwidth [143]. A defected structure can effectively reduce surface waves and decrease cross-polarization levels by confining surface waves within the dielectric that causes coupling between antenna radiating elements [144], [145].



FIGURE 6. Key design parameters of antenna arrays and mutual coupling reduction techniques.

The surface wave reduction results in less diffraction from the substrate which leads to a reduction in back radiation [146]. DGS can provide low mutual coupling, maximum efficiency and wide bandwidth to the antenna design.

2) ELECTROMAGNETIC BANDGAP (EBG) STRUCTURE

Electromagnetic Band Gap (EBG) structures are periodic structures that could allow or prevent the propagation of electromagnetic (EM) waves in some frequency bands over certain incident angles and polarization states [147]. These frequencies are called partial band-gap. EBG structure is a periodic arrangement of metallic or dielectric material. The periodicity of the EBG structure can generate individual resonance of elements that produce multiple band gaps [148]. At a specific frequency band, EBG does not allow the propagation of EM waves in all directions and incident angles and this frequency band is called a global or complete bandgap [149]. EBG structure is useful to suppress surface waves, which results in more efficient antennas with reduced coupling. EBG structures can effectively increase operation bandwidth, directivity, and front-to-back ratio and reduce side-lobe levels. EBG structures provide miniaturization, low mutual coupling and high efficiency.

3) METAMATERIALS

Metamaterials are artificial periodic or aperiodic structures that can demonstrate uncommon and exotic electromagnetic properties like negative permittivity and permeability [150]. Properties of metamaterials are determined by the periodic arrangement of scattering structures that are much smaller than the wavelength of EM waves. These small structured metamaterials can vary in size, orientation, and geometry and are generally fabricated from plastic and metal [151]. Antennas designed with metamaterials have significantly high radiation power. Antenna systems with embedded metamaterials can manipulate EM waves in a different way than conventional materials. Integration of metamaterials in antennas can provide high flexibility and design novelty. Metamaterials can offer miniaturization, high gain, low mutual coupling and high bandwidth.

4) DECOUPLING NETWORKS

The decoupling antenna system is realized by adding transmission lines or discrete components. In a decoupling network, admittance is transformed to imaginary value by the addition of discrete components [152], [153], coupled resonators [154], eigenmode decomposition [155], and artificial structure [156] or dummy load. Decoupling networks can minimize the effects of mutual coupling in antenna arrays at either single or dual-frequency. Dual-frequency decoupling can be achieved by using a network of series or parallel resonant components [157]. Decoupling networks provide high diversity gain, low mutual coupling and high isolation.

5) SLOT OR PARASITIC ELEMENTS

In this technique, slots or parasitic elements are added to the ground plane or radiation path that enhance the impedance bandwidth of the antenna system. For patch antennas, the current distribution is not uniform and areas with low current density can be removed as it has no significant effect on the return loss. Implementation of slots can increase the bandwidth and dimensions of the antenna and could suppress low current density areas [158]. Slot antennas can provide high gain, wide bandwidth and high efficiency. Parasitic elements can increase the gain and bandwidth of the patch antenna. Surface wave losses can also be reduced using parasitic patches, which results in mutual coupling reduction for patch arrays [159].

6) NEUTRALIZATION LINES

Neutralization lines are employed to propagate EM waves from one antenna to the other through the lumped element or metallic slit. This creates an opposite coupling to decrease the mutual coupling at specific frequencies between the antennas. This technique reduces the overall antenna size, improves bandwidth and reduces mutual coupling between antenna elements. Thin printed neutralization lines have been used to increase the isolation between two monopoles [160]. A thin neutralization line is etched near the antenna feeding line to link the monopoles. By varying the length of the neutralization line, isolation and impedance bandwidth was controlled.

Pair of crossed neutralization lines were used to mitigate mutual coupling for a dual-band antenna with two symmetric antenna elements [161]. Isolation between antenna elements was enhanced through crossed neutralization lines, driven branch, vias and parasitic ground. Pair of ground plane neutralization lines were utilized to reduce mutual coupling between four antenna elements placed orthogonal to each other [162]. A partial ground plane with rectangular slots and twin neutralization lines helped to operate the MIMO antenna system on multiple frequency bands with enhanced isolation.

7) DIELECTRIC RESONATOR

Dielectric resonator antennas (DRA) are designed using ceramic blocks that are integrated into the ground plane or metal surface [163], [164]. DRA can transform guided signals into unguided RF waves for propagation through water, air or vacuum. DRA has a much wider impedance bandwidth as compared to the microstrip antenna. This is due to the radiation of DRA through the complete DRA surface except for the grounded part, compared to the microstrip radiation via two narrow radiation slots. Different techniques can be used to excite dielectric resonators which include probe, aperture and microstrip line and these can be either singlepoint, multiplepoint or sequential rotation.

DRAs are designed in a circular, cylindrical [165], hemispherical [166], conical, triangular [167], trapezoidal [168] and rectangular [169] with cylindrical being the most common due to higher efficiency and lower loss compared to metal-only antennas. The size of DRA depends on the resonant frequency and dielectric constant of the materials. DRA provide high gain, low loss, high radiation efficiency and high isolation.

8) FREQUENCY RECONFIGURATION

Frequency reconfiguration is used to switch the frequency of operation to the desired range. This reconfiguration is realized by using varactor diodes, P-I-N diodes or Micro- Electro-Mechanical-Systems (MEMS) switches [170]. Reconfigurable antennas are designed to change the radiation parameters of an antenna which include frequency, radiation pattern, polarization or a combination of these parameters [171]. This antenna reconfiguration is realized by changing the feed location of the antenna or switching ground or radiating path that changes the corresponding resonating length. The controller circuit can be either microcontroller, Arduino or FPGA [172].

The two types of reconfigurable antennas are mechanical and electronically reconfigurable antennas. Electronic reconfiguration is achieved by switching the path of surface current on the ground or radiating element of an antenna. One or more switches can be used to switch paths that can be either P-I-N diodes, varactor diodes or MEMS switches [173], [174]. Schottky diode is used for high-speed switching and varactor diodes provide fine-tuning of frequency within the required band [175]. Mechanical reconfiguration is achieved by conductive fluid antennas, dielectric fluid antennas, gearing and origami-based antennas [172]. Reconfigurable antenna systems are capable of providing high diversity gain, low mutual coupling and high efficiency.

F. ENSURE COST-EFFECTIVE, PORTABLE AND LOW-POWER SOLUTIONS

Wearable gadgets have recently gained much attention due to their wide range of applications in sports, navigation, military, medical and space industry. A major bottleneck in the optimization of electronic components for wearable applications is antenna design. These antennas must either be built into the human body, like wearables or incorporated within the clothing. They must be low-cost, low-profile, portable, resilient, reliable and low-power, else they will be ineffective for portable applications.

Radiofrequency systems are low cost with low power requirements and microwave imaging produces non-ionized radiations which makes them more effective than other medical imaging techniques [176]. Compared to the complex traditional medical equipment with high maintenance and implementation costs, microwave systems are portable and consist of a microwave source, receiver, antenna array and radiofrequency switch to shift between the antenna elements. For wearable microwave devices, integration of Vector Network Analyzer (VNA) and radiofrequency switches in the device can make it more portable, convenient and less obtrusive. Several design challenges are involved as the wearable antennas are subjected to the near-field effects of the body, channel interference and user acceptance problems [177]. To increase the battery life of the wearable microwave system, the energy efficiency of the device should be improved. RF transmission by the antenna sensors constitutes a major part of the overall energy consumption. This problem can be mitigated by decreasing the number of retransmissions and improving the link budget through increasing antenna diversity gain.

G. MITIGATION OF NEAR-BODY EFFECTS ON PORTABLE SENSOR SYSTEMS AND SPECIFIC ABSORPTION RATE (SAR)

Microwave systems designed for wearable applications usually operate within proximity to the human body. This not only has an impact on the performance of sensors but can also cause damage to human health through electromagnetic radiation. Longer exposure to electromagnetic radiation can result in adverse reactions in the central nervous system which can lead to cognitive impairment and pose a higher risk of brain tumours [178]. There are some beneficial effects of short-term exposure to microwave radiation, it can improve cognitive functions, short-term memory loss and attention disorder [179]. Moreover, the risk of Alzheimer's disease is around 30 percent lower in people who use a mobile phone regularly for more than 10 years compared to other individuals [180]. The radiation intensity is a major factor to consider while designing wearable antennas. SAR is a criterion for determining if an antenna is safe for the human body, and it refers to the ratio of radiation generated by the antenna to the amount of absorption. There are two standards in place, the IEEE standard allows maximum absorption of 1.6W/Kg for 1g and International Commission for Non-Ionizing Radiation Protection permits a maximum of 2W/Kg per 10g [181].

X. FUTURE RESEARCH

Microwave imaging and sensing techniques are promising alternatives to conventional imaging technologies. Portable microwave devices can provide a low-cost and robust detection for the majority of diseases through wearable applications. Integration of these portable devices with cloud databases will be beneficial for post-processing and diagnosis using cloud analytics. Ongoing future work, as part of the COG-MHEAR project [182], aims to develop and evaluate the use of emerging portable non-invasive sensing technologies [183] for the detection of cognitive load within a hearing aid device. This device could be used by people with hearing loss in smart care home settings to enhance their quality of life. Other aspects and research areas that can be explored in future are discussed below:

A. UNIFIED DEVICE FOR HEARING LOSS, COGNITIVE LOAD AND DEMENTIA MONITORING AND DETECTION

The possibility of an integrated device which can work in conjunction with the hearing aid module needs to be explored. Microwave portable non-invasive sensing technologies can be used in smart care home settings for the detection of cognitive load within a hearing aid device. This can be beneficial to capture the progression of neurodegeneration at a very early stage. Furthermore, these devices can be made easily available to smart care homes and regular monitoring can be ensured. This will lead to ease of burden on medical facilities and clinicians, as the initial diagnosis can be easily made without much clinical intervention.

B. CLINICIAN'S PARTICIPATION IN IMPROVEMENT OF DESIGNED SOLUTION

The involvement of clinicians in the design of microwave sensing systems can make the solution more practical and implementable in medical and care home setups. As the medical practitioners are more knowledgeable about the sensitivity of patients with medical devices, they can provide valuable inputs to make the devices user-friendly, safe and user-centric. Furthermore, they are in a better position to compare these portable sensing devices with conventional equipment to highlight the shortcomings and mitigate the challenges. Other than that, their feedback can help in the improvement of imaging techniques during the development phase.

C. PRIVACY AND SECURITY MANAGEMENT

Another challenge in widespread portable microwave sensors is to maintain the privacy of the patients. As these devices can be integrated with Wi-Fi to store the patient's data on the cloud as a medical record. The integrity and security of the data storage systems must be maintained to avoid any misuse and leakage of patient data. This requires viable solutions and data protection policies to be implemented in

future for the maintenance of the data privacy rights of the patients involved. Homomorphic encryption can be used to maintain data privacy while performing data computations in the cloud [184]. It provides end-to-end dataflow privacy and enables secure storage of data in public clouds. This helps in data computations and Machine Learning (ML) predictions without accessing user data and decryption. Although it increases computational cost and packet size but helps to maintain the redundancy efficiency of distributed storage [185]. Federated learning is another privacy learning ML algorithm which enables computation and insights without moving patient data beyond the firewalls of their residing institution [186]. It enables multiple stakeholders to train collaboratively without the need to exchange or centralize the data sets. Federated learning can achieve adequate differential privacy and avoid sharing healthcare data and information leakage [187]. Blockchain is another emerging technology which can provide data privacy and security for patient data and clinical information. The blockchain model allows building a model through patients record on a distributed ledger where all members contribute to the consistency and integrity of the database [188]. A blockchain-based system can keep a record of not only original data but also holds information of all transformation applied to the data, that leads to quick and efficient detection of falsified figures [189].

D. MACHINE LEARNING BASED MICROWAVE SENSING AND IMAGING ALGORITHMS

In future, portable microwave devices can be linked to the cloud and the data obtained can be post-processed with cloud analytics and machine learning for better diagnostics and care services. The scattering parameters data received from the antenna sensors can be stored and compared with the output of conventional techniques to create suitable data for the machine intelligence system. This data can then be post-processed using suitable supervised machine learning algorithms like Nearest Neighbour (NN) [190], k-Nearest Neighbour (kNN) [191], and Support Vector Machine (SVM) [192] or Multi-Layer Perceptron (MLP) neural network [193], [194].

The selection of an imaging algorithm is crucial for accurate diagnosis and imaging of different neurodegenerative diseases. Imaging algorithms presented in the recent literature can generate a 2D image of the brain, but the exact location of a tumour or affected brain can be more accurately determined by 3D imaging. Therefore, future research on advanced 3D microwave imaging algorithms can improve the diagnosis. Microwave imaging can be enhanced by the adaptation of techniques like Machine Learning [195], [196], Artificial Intelligence and Big Data [197] to consolidate data from various use case, patients and their disease progression levels. This data can be used to compare the diagnosis of the target patient and accurately reconcile the results. Recent research has been focused on utilizing a deep learning network [198], PySpark-based reconstruction [199] and parallel delay multiply and sum image reconstruction algorithm [200], which performed more efficiently than the delay multiple and sum algorithm. The experimental results from these researches have shown that the Spark significantly accelerates the imaging reconstruction process without affecting the accuracy.

E. REALISTIC PHANTOMS DEVELOPMENT

Another challenge is to emulate the neurodegenerative diseases with realistic phantom models. This requires an initial dielectric measurement of the brain tissues, grey matter, white matter, blood pool and cerebrospinal fluid. Different layers can be used to realize the actual brain with the flexibility to control dielectric properties through slots, channels and pumps. Realistic phantoms can be designed using various techniques. Liquid phantoms are reported in [201] for the study of electromagnetic hyperthermia. This liquid phantom is composed of using a mixture of sodium chloride, sucrose and water. Liquid phantoms are the most flexible and commonly adopted models in biomedical microwave sensing. In another study, a gel-like time-durable phantom is presented to mimic skin, muscle, fat and blood [202]. This model is useful for the prediction of bio-effects at microwave and millimetre-wave frequencies. Polymer composition materials are used to fabricate stable and long-life head phantoms [203]. Polyepoxides (Epoxy), graphite, aluminium oxide, carbon and brass powders are used to fabricate the grey and white matter part of the brain. Blood mimicking material is used to represent stroke conditions and waterbased material is utilized to emulate cerebrospinal fluid. Compared to liquid and gel-like phantoms, 3D printable solid phantoms can provide more realistic, stable and robust phantoms. The only drawback of these 3D-printed phantoms is the additive manufacturing and availability of materials [204]. These issues will certainly be resolved with the advancements in 3D printing technology as it will increasingly allow more materials to be processed in near future.

XI. CONCLUSION

Non-invasive portable sensors are being developed for conditions such as dementia and neurodegeneration detection. The integration of these sensors with a hearing aid device could lead to more effective procedures for the detection and treatment of more complex conditions arising in care settings. The current study presents an overview of the interrelation between hearing loss, cognitive load and dementia. Current research trends related to monitoring and detection of cognitive load are discussed, and the possibility of cognitive load detection through RF and microwave sensors is discussed in conjunction with hearing aid devices. Practical implications and challenges that can be faced in the realization of these devices are discussed, along with the possible viable solutions from existing literature. RF and microwave portable technology has the potential to revolutionize care homes and diagnostic radiology setups and to replace conventional diagnostic equipment with portable, low cost and non-invasive wearable devices. Integration with cloud computation and machine learning can further optimize the imaging analysis,

diagnosis and prognosis through microwave sensors. Hence, this portable microwave technology holds immense potential if the discussed challenges and implications will be mitigated.

REFERENCES

- [1] A. Davis, C. M. Mcmahon, K. M. Pichora-Fuller, S. Russ, F. Lin, B. O. Olusanya, S. Chadha, and K. L. Tremblay, "Aging and hearing health: The life-course approach," *Gerontologist*, vol. 56, no. 2, pp. S256–S267, Apr. 2016, doi: 10.1093/geront/gnw033.
- [2] G. Easton and T. Leverton, "Supporting adults with hearing loss in primary care: New Nice guideline," *Brit. J. Gen. Pract.*, vol. 68, no. 676, pp. 516–517, Nov. 2018, doi: 10.3399/bjgp18X699497.
- [3] M. Ray, T. Dening, and B. Crosbie, "Dementia and hearing loss: A narrative review," *Maturitas*, vol. 128, pp. 64–69, Oct. 2019, doi: 10.1016/j.maturitas.2019.08.001.
- [4] P. J. Lundine and J. R. McCauley, "A tutorial on expository discourse: Structure, development, and disorders in children and adolescents," *Amer. J. Speech-Lang. Pathol.*, vol. 25, pp. 1–15, Oct. 2016, doi: 10.1044/2016_AJSLP-14-0130.
- [5] Research Impact Report 2014, C. C. S. R. Inst., Brit. Acad. Audiology, Scotland, U.K., May 2014, pp. 1–8. [Online]. Available: http://www.baaudiology.org
- [6] B. C. M. Stephan, T. Minett, E. Pagett, M. Siervo, C. Brayne, and I. G. McKeith, "Diagnosing mild cognitive impairment (MCI) in clinical trials: A systematic review," *BMJ Open*, vol. 3, no. 2, pp. 1–8, 2013, doi: 10.1136/bmjopen-2012-001909.
- [7] M. J. Angevaare, J. M. J. Vonk, L. Bertola, L. Zahodne, C. W.-M. Watson, A. Boehme, N. Schupf, R. Mayeux, M. I. Geerlings, and J. J. Manly, "Predictors of incident mild cognitive impairment and its course in a diverse community-based population," *Neurology*, vol. 98, no. 1, pp. e15–e26, Jan. 2022, doi: 10.1212/WNL.000000 0000013017.
- [8] C. Iadecola, K. Yaffe, J. Biller, L. C. Bratzke, F. M. Faraci, P. B. Gorelick, M. Gulati, H. Kamel, D. S. Knopman, L. J. Launer, J. S. Saczynski, S. Seshadri, and A. Z. Al Hazzouri, "Impact of hypertension on cognitive function: A scientific statement from the American heart association," *Hypertension*, vol. 68, no. 6, Dec. 2016, doi: 10.1161/HYP.000000000000053.
- [9] J. Rasmussen and H. Langerman, "Alzheimer's disease—Why we need early diagnosis," *Degenerative Neurol. Neuromuscular Disease*, vol. 9, pp. 123–130, Dec. 2019, doi: 10.2147/dnnd.s228939.
- [10] I. Saied, T. Arslan, S. Chandran, C. Smith, T. Spires-Jones, and S. Pal, "Non-invasive RF technique for detecting different stages of Alzheimer's disease and imaging beta-amyloid plaques and tau tangles in the brain," *IEEE Trans. Med. Imag.*, vol. 39, no. 12, pp. 4060–4070, Dec. 2020, doi: 10.1109/TMI.2020.3011359.
- [11] I. M. Saied and T. Arslan, "Noninvasive wearable RF device towards monitoring brain atrophy and lateral ventricle enlargement," *IEEE J. Electromagn., RF Microw. Med. Biol.*, vol. 4, no. 1, pp. 61–68, Mar. 2020, doi: 10.1109/JERM.2019.2926163.
- [12] S. A. Rezaeieh, A. Zamani, and A. M. Abbosh, "3-D wideband antenna for head-imaging system with performance verification in brain tumor detection," *IEEE Antennas Wireless Propag. Lett.*, vol. 14, pp. 910–914, 2015, doi: 10.1109/LAWP.2014.2386852.
- [13] A. S. M. Alqadami, A. Trakic, A. E. Stancombe, B. Mohammed, K. Bialkowski, and A. Abbosh, "Flexible electromagnetic cap for head imaging," *IEEE Trans. Biomed. Circuits Syst.*, vol. 14, no. 5, pp. 1097–1107, Oct. 2020, doi: 10.1109/TBCAS.2020.3025341.
- [14] M. T. Islam, M. Z. Mahmud, M. T. Islam, S. Kibria, and M. Samsuzzaman, "A low cost and portable microwave imaging system for breast tumor detection using UWB directional antenna array," *Sci. Rep.*, vol. 9, no. 1, pp. 1–13, Dec. 2019, doi: 10.1038/s41598-019-51620-z.
- [15] C. J. D. Hardy, C. R. Marshall, H. L. Golden, C. N. Clark, C. J. Mummery, T. D. Griffiths, D.-E. Bamiou, and J. D. Warren, "Hearing and dementia," *J. Neurol.*, vol. 263, no. 11, pp. 2339–2354, Nov. 2016, doi: 10.1007/s00415-016-8208-y.
- [16] T. D. Griffiths, M. Lad, S. Kumar, E. Holmes, B. Mcmurray, E. A. Maguire, A. J. Billig, and W. Sedley, "How can hearing loss cause dementia?" *Neuron*, vol. 108, no. 3, pp. 401–412, Nov. 2020, doi: 10.1016/j.neuron.2020.08.003.

- [17] W. Xu, C. Zhang, J.-Q. Li, C.-C. Tan, X.-P. Cao, L. Tan, J.-T. Yu, and A. Disease Neuroimaging Initiative, "Age-related hearing loss accelerates cerebrospinal fluid tau levels and brain atrophy: A longitudinal study," *Aging*, vol. 11, no. 10, pp. 3156–3169, May 2019, doi: 10.18632/aging.101971.
- [18] S. Fortunato, F. Forli, V. Guglielmi, E. De Corso, G. Paludetti, S. Berrettini, and A. R. FetonI, "Ipoacusia e declino cognitivo: Revisione della letteratura," *Acta Otorhinolaryngologica Italica*, vol. 36, no. 3, pp. 155–166, May 2016, doi: 10.14639/0392-100X-993.
- [19] A. Ciorba, C. Bianchini, S. Pelucchi, and A. Pastore, "The impact of hearing loss on the quality of life of elderly adults," *Clin. Interv. Aging*, vol. 7, pp. 159–163, Jun. 2012, doi: 10.2147/CIA.S26059.
- [20] J. Wei, Y. Hu, L. Zhang, Q. Hao, R. Yang, H. Lu, X. Zhang, and E. K. Chandrasekar, "Hearing impairment, mild cognitive impairment, and dementia: A meta-analysis of cohort studies," *Dementia Geriatric Cognit. Disorders Extra*, vol. 7, no. 3, pp. 440–452, Dec. 2017, doi: 10.1159/000485178.
- [21] S. Li, C. Cheng, L. Lu, X. Ma, X. Zhang, A. Li, J. Chen, X. Qian, and X. Gao, "Hearing loss in neurological disorders," *Frontiers Cell Develop. Biol.*, vol. 9, pp. 1–16, Aug. 2021, doi: 10.3389/fcell. 2021.716300.
- [22] J. C. S. Johnson, C. R. Marshall, R. S. Weil, D.-E. Bamiou, C. J. D. Hardy, and J. D. Warren, "Hearing and dementia: From ears to brain," *Brain*, vol. 144, no. 2, pp. 391–401, Mar. 2021, doi: 10.1093/brain/awaa429.
- [23] R. R.-K. Benjamin, D. Granta, C. A. Smithb, P. E. Castlec, and M. E. Scheurere, "HHS public access," *Physiol. Behav.*, vol. 176, no. 5, pp. 139–148, 2017, doi: 10.1016/j.heares.2016.05.018.
- [24] Y. Uchida, S. Sugiura, Y. Nishita, N. Saji, M. Sone, and H. Ueda, "Agerelated hearing loss and cognitive decline—The potential mechanisms linking the two," *Auris Nasus Larynx*, vol. 46, no. 1, pp. 1–9, Feb. 2019, doi: 10.1016/j.anl.2018.08.010.
- [25] D. P. Phillips, "Central auditory system and central auditory processing disorders: Some conceptual issues," *Semin. Hear.*, vol. 23, no. 4, pp. 251–261, 2002, doi: 10.1055/s-2002-35875.
- [26] P. A. Tun, S. McCoy, and A. Wingfield, "Aging, hearing acuity, and the attentional costs of effortful listening," *Psychol. Aging*, vol. 24, no. 3, pp. 761–766, 2009, doi: 10.1037/a0014802.
- [27] T. Si, G. Xing, and Y. Han, "Subjective cognitive decline and related cognitive deficits," *Frontiers Neurol.*, vol. 11, pp. 1–13, May 2020, doi: 10.3389/fneur.2020.00247.
- [28] A. K. Gibson and V. E. Richardson, "Living alone with cognitive impairment: Findings from the national health and aging trends study," *Amer. J. Alzheimer's Disease Dementiasr*, vol. 32, no. 1, pp. 56–62, Feb. 2017, doi: 10.1177/1533317516673154.
- [29] S. Castles, World Population Movements, Diversity, and Education, vol. 43, no. 2. New York, NY, USA: Routledge, 2011.
- [30] S.-Y. Tai, C.-T. Shen, L.-F. Wang, and C.-Y. Chien, "Association of sudden sensorineural hearing loss with dementia: A nationwide cohort study," *BMC Neurol.*, vol. 21, no. 1, pp. 1–8, Dec. 2021, doi: 10.1186/s12883-021-02106-x.
- [31] J. Singer, J. N. Trollor, B. T. Baune, P. S. Sachdev, and E. Smith, "Arterial stiffness, the brain and cognition: A systematic review," *Ageing Res. Rev.*, vol. 15, pp. 16–27, May 2014, doi: 10.1016/j.arr. 2014.02.002.
- [32] Z. Ungvari, P. Toth, S. Tarantini, C. I. Prodan, F. Sorond, B. Merkely, and A. Csiszar, "Hypertension-induced cognitive impairment: From pathophysiology to public health," *Nature Rev. Nephrol.*, vol. 17, no. 10, pp. 639–654, Oct. 2021, doi: 10.1038/s41581-021-00430-6.
- [33] S. Shityakov, K. Hayashi, S. Störk, V. Scheper, T. Lenarz, and C. Y. Förster, "The conspicuous link between ear, brain and heartcould neurotrophin-treatment of age-related hearing loss help prevent Alzheimer's disease and associated amyloid cardiomyopathy?" *Biomolecules*, vol. 11, no. 6, p. 900, Jun. 2021, doi: 10.3390/ biom11060900.
- [34] S. Ahn, M. A. Mathiason, D. Salisbury, and F. Yu, "Factors predicting the onset of amnestic mild cognitive impairment or Alzheimer's dementia in persons with subjective cognitive decline," *J. Gerontol. Nursing*, vol. 46, no. 8, pp. 28–36, Aug. 2020, doi: 10.3928/00989134-20200619-01.
- [35] J. L. Bryant, J. D. Boughter, G. Suzhen, M. S. LeDoux, and D. H. Heck, "HHS public access," *Physiol. Behav.*, vol. 32, no. 1, pp. 41–52, 2010, doi: 10.1016/j.cger.2013.07.009.

- [36] "Annual facts and figures report by Alzheimer's association," *Alzheimer's Dement.*, vol. 15, no. 3, pp. 321–387, 2019, doi: 10.1016/j.jalz.2019.01.010.
- [37] Addressing The Rising Prevalence of Hearing Loss, no. 2, OMS, World Health Org., Geneva, Switzerland, 2018.
- [38] E. Brai, A. Tonacci, V. Brugada-Ramentol, F. D'Andrea, and L. Alberi, "Intercepting dementia: Awareness and innovation as key tools," *Frontiers Aging Neurosci.*, vol. 13, pp. 1–8, Oct. 2021, doi: 10.3389/fnagi.2021.730727.
- [39] A. R. Fetoni, A. Pisani, R. Rolesi, F. Paciello, A. Viziano, A. Moleti, R. Sisto, D. Troiani, G. Paludetti, and C. Grassi, "Early noise-induced hearing loss accelerates presbycusis altering aging processes in the cochlea," *Frontiers Aging Neurosci.*, vol. 14, p. 12, Feb. 2022, doi: 10.3389/fnagi.2022.803973.
- [40] A. Martini, A. Castiglione, R. Bovo, A. Vallesi, and C. Gabelli, "Aging, cognitive load, dementia and hearing loss," *Audiol. Neurotol.*, vol. 19, no. 1, pp. 2–5, 2014, doi: 10.1159/000371593.
- [41] K. Lu, J. M. Nicholas, J. D. Collins, S.-N. James, T. D. Parker, C. A. Lane, A. Keshavan, S. E. Keuss, S. M. Buchanan, H. Murray-Smith, D. M. Cash, C. H. Sudre, I. B. Malone, W. Coath, A. Wong, S. M. D. Henley, S. J. Crutch, N. C. Fox, M. Richards, and J. M. Schott, "Cognition at age 70: Life course predictors and associations with brain pathologies," *Neurology*, vol. 93, no. 23, pp. e2144–e2156, Dec. 2019, doi: 10.1212/WNL.000000000008534.
- [42] K. Slade, C. J. Plack, and H. E. Nuttall, "The effects of age-related hearing loss on the brain and cognitive function," *Trends Neurosci.*, vol. 43, no. 10, pp. 810–821, Oct. 2020, doi: 10.1016/j.tins.2020.07.005.
- [43] H.-C. Lin, P.-Z. Chao, and H.-C. Lee, "Sudden sensorineural hearing loss increases the risk of stroke: A 5-year follow-up study," *Stroke*, vol. 39, no. 10, pp. 2744–2748, Oct. 2008, doi: 10.1161/STROKEAHA.108.519090.
- [44] M. Khosravipour and F. Rajati, "Sensorineural hearing loss and risk of stroke: A systematic review and meta-analysis," *Sci. Rep.*, vol. 11, no. 1, pp. 1–11, Dec. 2021, doi: 10.1038/s41598-021-89695-2.
- [45] N. M. Armstrong, O. A. Williams, B. A. Landman, J. A. Deal, F. R. Lin, and S. M. Resnick, "Association of poorer hearing with longitudinal change in cerebral white matter microstructure," *JAMA Otolaryngol. Head Neck Surg.*, vol. 146, no. 11, pp. 1035–1042, 2020, doi: 10.1001/jamaoto.2020.2497.
- [46] D. M. P. Jayakody, P. L. Friedland, R. N. Martins, and H. R. Sohrabi, "Impact of aging on the auditory system and related cognitive functions: A narrative review," *Frontiers Neurosci.*, vol. 12, pp. 1–16, Mar. 2018, doi: 10.3389/fnins.2018.00125.
- [47] Z. J. Qian, P. D. Chang, G. Moonis, and A. K. Lalwani, "A novel method of quantifying brain atrophy associated with age-related hearing loss," *NeuroImage, Clin.*, vol. 16, pp. 205–209, Jan. 2017, doi: 10.1016/j.nicl.2017.07.021.
- [48] N. Giroud, M. K. Pichora-Fuller, P. Mick, W. Wittich, F. Al-Yawer, S. Rehan, J. B. Orange, and N. A. Phillips, "Hearing loss is associated with gray matter differences in older adults at risk for and with Alzheimer's disease," *Aging Brain*, vol. 1, 2021, Art. no. 100018, doi: 10.1016/j.nbas.2021.100018.
- [49] J. E. Peelle, V. Troiani, M. Grossman, and A. Wingfield, "Hearing loss in older adults affects neural systems supporting speech comprehension," *J. Neurosci.*, vol. 31, no. 35, pp. 12638–12643, Aug. 2011, doi: 10.1523/JNEUROSCI.2559-11.2011.
- [50] H. Y. Tarawneh, H. K. Menegola, A. Peou, H. Tarawneh, and D. M. P. Jayakody, "Central auditory functions of Alzheimer's disease and its preclinical stages: A systematic review and meta-analysis," *Cells*, vol. 11, no. 6, pp. 1–24, 2022, doi: 10.3390/cells11061007.
- [51] J. Häggström, U. Rosenhall, C. Hederstierna, P. Östberg, and E. Idrizbegovic, "A longitudinal study of peripheral and central auditory function in Alzheimer's disease and in mild cognitive impairment," *Dementia Geriatric Cognit. Disorders Extra*, vol. 8, no. 3, pp. 393–401, Oct. 2018, doi: 10.1159/000493340.
- [52] G. Mertens, E. Andries, A. J. Claes, V. Topsakal, P. Van de Heyning, V. Van Rompaey, M. Calvino, I. S. Cuadrado, E. Muñoz, J. Gavilán, and K. Bienkowska, "Cognitive improvement after cochlear implantation in older adults with severe or profound hearing impairment: A prospective, longitudinal, controlled, multicenter study," *Ear Hearing*, vol. 42, no. 3, pp. 606–614, 2021, doi: 10.1097/AUD.00000000000962.
- [53] B. C.-K. Tong, "HHS public access," *Physiol. Behav.*, vol. 176, no. 5, pp. 139–148, 2017, doi: 10.1016/j.jagp.2016.08.019.

- [54] J. Sarant, D. Harris, P. Busby, P. Maruff, A. Schembri, U. Lemke, and S. Launer, "The effect of hearing aid use on cognition in older adults: Can we delay decline or even improve cognitive function?" *J. Clin. Med.*, vol. 9, no. 1, p. 254, Jan. 2020, doi: 10.3390/jcm9010254.
- [55] A. Chern, J. S. Golub, and A. K. Lalwani, "Do hearing aids help prevent cognitive decline?" *Laryngoscope*, vol. 131, no. 10, pp. 2166–2168, Oct. 2021, doi: 10.1002/lary.29365.
- [56] H. Amieva and C. Ouvrard, "Does treating hearing loss in older adults improve cognitive outcomes? A review," J. Clin. Med., vol. 9, no. 3, p. 805, Mar. 2020, doi: 10.3390/jcm9030805.
- [57] M. Bucholc, P. L. McClean, S. Bauermeister, S. Todd, X. Ding, Q. Ye, D. Wang, W. Huang, and L. P. Maguire, "Association of the use of hearing aids with the conversion from mild cognitive impairment to dementia and progression of dementia: A longitudinal retrospective study," *Alzheimer's Dementia, Transl. Res. Clin. Interventions*, vol. 7, no. 1, pp. 1–11, Jan. 2021, doi: 10.1002/trc2.12122.
- [58] C. Pettigrew and A. Soldan, "Defining cognitive reserve and implications for cognitive aging," *Current Neurol. Neurosci. Rep.*, vol. 19, no. 1, pp. 1–22, Jan. 2019, doi: 10.1007/s11910-019-0917-z.
- [59] C. M. Liu and C. T. C. Lee, "Association of hearing loss with dementia," JAMA Netw. Open, vol. 2, no. 7, pp. 1–15, 2019, doi: 10.1001/jamanetworkopen.2019.8112.
- [60] A. Martini, A. Castiglione, R. Bovo, A. Vallesi, and C. Gabelli, "Aging, cognitive load, dementia and hearing loss," *Audiol. Neurotol.*, vol. 19, no. Suppl. 1, pp. 2–5, 2014, doi: 10.1159/000371593.
- [61] J. Feldman and I. Barshi, "The effects of blood glucose levels on cognitive performance: A review of the literature," NASA Tech. Memorandum 2007-214555, Sacramento, CA, USA, Tech. Rep. NASA/TM-2007-214555, Jun. 2007.
- [62] F. J. Wolters, H. I. Zonneveld, A. Hofman, A. van der Lugt, P. J. Koudstaal, M. W. Vernooij, and M. A. Ikram, "Cerebral perfusion and the risk of dementia: A population-based study," *Circulation*, vol. 136, no. 8, pp. 719–728, Aug. 2017, doi: 10.1161/CIRCULATIONAHA. 117.027448.
- [63] J. J. Snodgrass, W. R. Leonard, M. V. Sorensen, L. A. Tarskaia, and M. J. Mosher, "The influence of basal metabolic rate on blood pressure among indigenous siberians," *Amer. J. Phys. Anthropol.*, vol. 137, no. 2, pp. 145–155, Oct. 2008, doi: 10.1002/ajpa.20851.
- [64] J. Moini and P. Piran, "Histophysiology," in *Functional and Clinical Neuroanatomy*. Amsterdam, The Netherlands: Elsevier, 2020.
- [65] B. C.-K. Tong, "HHS public access," *Physiol. Behav.*, vol. 176, no. 5, pp. 139–148, 2017, doi: 10.1016/j.it.2008.05.002.
- [66] Public Health Response to Dementia, World Health Org., Geneva, Switzerland, 2021.
- [67] "Annual facts and figures report by Alzheimer's association," *Alzheimer's Dement.*, vol. 17, no. 3, pp. 327–406, 2021, doi: 10.1002/alz.12328.
- [68] M. Guerchet, M. Prince, and M. Prina. (2020). Numbers of People With Dementia Around the World. [Online]. Available: https://www.alzint.org/resource/numbers-of-people-with-dementiaworldwide/
- [69] (2017). W. H. Organisation. Global Action Plan on the Public Health Response to Dementia 2017–2025. [Online]. Available: http://www.who. int/mental_health/neurology/dementia/action_plan_2017_2025/en/
- [70] (2021). U.K. Official Statistics-Dementia Profile, March 2021. [Online]. Available: https://www.gov.uk/government/statistics/dementia-profileupdates/statistical-commentary-dementia-profile-march-2021-update
- [71] (2022). U.K. Official Statistics-Dementia Diagnoses, January 2022.
 [Online]. Available: https://digital.nhs.uk/data-and-information/ publications/statistical/recorded-dementia-diagnoses/january-2022.
- [72] "Annual facts and figures report by Alzheimer's association," *Alzheimer's Dement.*, vol. 16, no. 3, pp. 391–460, 2020, doi: 10.1002/alz.12068.
- [73] G. S. Bloom, "Amyloid-β and tau: The trigger and bullet in Alzheimer disease pathogenesis," *JAMA Neurol.*, vol. 71, no. 4, pp. 505–508, 2014, doi: 10.1001/jamaneurol.2013.5847.
- [74] K. Sharma, S. Pradhan, L. K. Duffy, S. Yeasmin, N. Bhattarai, and M. K. Schulte, "Role of receptors in relation to plaques and tangles in Alzheimer's disease pathology," *Int. J. Mol. Sci.*, vol. 22, no. 23, p. 12987, Nov. 2021, doi: 10.3390/ijms222312987.
- [75] S. L. Leal, S. M. Landau, R. K. Bell, and W. J. Jagust, "Hippocampal activation is associated with longitudinal amyloid accumulation and cognitive decline," *eLife*, vol. 6, pp. 1–15, Feb. 2017, doi: 10.7554/eLife.22978.

- [76] M. Kolarova, F. García-Sierra, A. Bartos, J. Ricny, and D. Ripova, "Structure and pathology of tau protein in Alzheimer disease," *Int. J. Alzheimer's Disease*, vol. 2012, pp. 1–13, Jan. 2012, doi: 10.1155/2012/731526.
- [77] F. J. Wolters and M. A. Ikram, "Epidemiology of vascular dementia: Nosology in a time of epiomics," *Arteriosclerosis, Thrombosis, Vascular Biol.*, vol. 39, no. 8, pp. 1542–1549, Aug. 2019, doi: 10.1161/ATVBAHA.119.311908.
- [78] S. Ray and S. Davidson, "Dementia and cognitive decline A review of the evidence," Age U.K., vol. 27, pp. 10–12, 2014.
- [79] A. F. Kurz, "What is vascular dementia?" Int. J. Clin. Pract. Suppl., vol. 120, no. 120, pp. 5–8, 2001.
- [80] E. McKay and S. E. Counts, "Multi-infarct dementia: A historical perspective," *Dementia Geriatric Cognit. Disorders Extra*, vol. 7, no. 1, pp. 160–171, May 2017, doi: 10.1159/000470836.
- [81] A. Ribeiro, O. Husson, N. Drey, I. Murray, K. May, J. Thurston, and W. Oyen, "Ionising radiation exposure from medical imaging—A review of patient's (un) awareness," *Radiography*, vol. 26, no. 2, pp. e25–e30, May 2020, doi: 10.1016/j.radi.2019.10.002.
- [82] T. Lauteslager, N. Nicolaou, T. S. Lande, and T. Constandinou, "Functional neuroimaging using UWB impulse radar: A feasibility study," in *Proc. IEEE Biomed. Circuits Syst. Conf. (BioCAS)*, Oct. 2015, pp. 1–4, doi: 10.1109/BioCAS.2015.7348387.
- [83] A. S. M. Alqadami, K. S. Bialkowski, A. T. Mobashsher, and A. M. Abbosh, "Wearable electromagnetic head imaging system using flexible wideband antenna array based on polymer technology for brain stroke diagnosis," *IEEE Trans. Biomed. Circuits Syst.*, vol. 13, no. 1, pp. 124–134, Oct. 2019, doi: 10.1109/TBCAS.2018.2878057.
- [84] Y. Shi, N. Ruiz, R. Taib, E. Choi, and F. Chen, "Galvanic skin response (GSR) as an index of cognitive load," in *Proc. CHI Extended Abstr. Hum. Factors Comput. Syst.*, Apr. 2007, pp. 2651–2656, doi: 10.1145/1240866.1241057.
- [85] V. Farrell, "SIGCHI (group: U.S.), human factors and ergonomic society of Australia. Computer-human interaction special interest group," *Proc.* 24th Austral. Comput.-Hum. Interact. Conf. (OzCHI), 2015, p. 692.
- [86] P. Weber, F. Rupprecht, S. Wiesen, B. Hamann, and A. Ebert, "Assessing cognitive load via pupillometry," in *Advances in Artificial Intelligence and Applied Cognitive Computing, Transactions on Computational Science and Computational Intelligence*. Cham, Switzerland: Springer, 2021, pp. 1087–1096, doi: 10.1007/978-3-030-70296-0_86.
- [87] M. Lohani, B. R. Payne, and D. L. Strayer, "A review of psychophysiological measures to assess cognitive states in real-world driving," *Frontiers Hum. Neurosci.*, vol. 13, pp. 1–27, Mar. 2019, doi: 10.3389/fnhum.2019.00057.
- [88] Z. T. Al-Sharify, T. A. Al-Sharify, N. T. Al-Sharify, and H. Y. Naser, "A critical review on medical imaging techniques (CT and PET scans) in the medical field," in *Proc. IOP Conf. Mater. Sci. Eng.*, vol. 870, no. 1, Jun. 2020, Art. no. 012043, doi: 10.1088/1757-899X/870/1/012043.
- [89] Usselman CWNSSJRB, "HHS public access," *Physiol. Behav.*, vol. 176, no. 3, pp. 139–148, 2017, doi: 10.1016/j.physbeh.2017.03.040.
- [90] K. Engedal, M. L. Barca, P. Høgh, B. B. Andersen, N. W. Dombernowsky, M. Naik, T. E. Gudmundsson, A.-R. Øksengaard, L.-O. Wahlund, and J. Snaedal, "The power of EEG to predict conversion from mild cognitive impairment and subjective cognitive decline to dementia," *Dementia Geriatric Cognit. Disorders*, vol. 49, no. 1, pp. 38–47, 2020, doi: 10.1159/000508392.
- [91] G. H. Glover, "Overview of functional magnetic resonance imaging," *Neurosurg. Clin.*, vol. 22, no. 2, pp. 133–139, 2011, doi: 10.1016/j.nec.2010.11.001.
- [92] D. McDuff, S. Gontarek, and R. Picard, "Remote measurement of cognitive stress via heart rate variability," in *Proc. 36th Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Aug. 2014, pp. 2957–2960, doi: 10.1109/EMBC.2014.6944243.
- [93] B. Mehler, B. Reimer, and Y. Wang, "A comparison of heart rate and heart rate variability indices in distinguishing single-task driving and driving under secondary cognitive workload," in *Proc. Driving Assessment Conf.*, 2011, pp. 590–597, doi: 10.17077/drivingassessment.1451.
- [94] Q. Wang, S. Yang, M. Liu, Z. Cao, and Q. Ma, "An eye-tracking study of website complexity from cognitive load perspective," *Decis. Support Syst.*, vol. 62, pp. 1–10, Jun. 2014, doi: 10.1016/j.dss.2014.02.007.
- [95] F. Amadieu, T. van Gog, F. Paas, A. Tricot, and C. Mariné, "Effects of prior knowledge and concept-map structure on disorientation, cognitive load, and learning," *Learn. Instruct.*, vol. 19, no. 5, pp. 376–386, Oct. 2009, doi: 10.1016/j.learninstruc.2009.02.005.

- [96] B. Mahanama, Y. Jayawardana, S. Rengarajan, G. Jayawardena, L. Chukoskie, J. Snider, and S. Jayarathna, "Eye movement and pupil measures: A review," *Frontiers Comput. Sci.*, vol. 3, pp. 1–22, Jan. 2022, doi: 10.3389/fcomp.2021.733531.
- [97] M. A. Hogervorst, A.-M. Brouwer, and J. B. F. van Erp, "Combining and comparing EEG, peripheral physiology and eye-related measures for the assessment of mental workload," *Frontiers Neurosci.*, vol. 8, pp. 1–14, Oct. 2014, doi: 10.3389/fnins.2014.00322.
- [98] M. D. Babu, D. V. JeevithaShree, G. Prabhakar, K. P. S. Saluja, A. Pashilkar, and P. Biswas, "Estimating pilots' cognitive load from ocular parameters through simulation and in-flight studies," *J. Eye Movement Res.*, vol. 12, no. 3, pp. 1–16, Sep. 2019, doi: 10.16910/jemr.12.3.3.
- [99] V. Borisov, E. Kasneci, and G. Kasneci, "Robust cognitive load detection from wrist-band sensors," *Comput. Hum. Behav. Rep.*, vol. 4, Aug. 2021, Art. no. 100116, doi: 10.1016/j.chbr.2021.100116.
- [100] R. R. Whelan, "Neuroimaging of cognitive load in instructional multimedia," *Educ. Res. Rev.*, vol. 2, no. 1, pp. 1–12, Jan. 2007, doi: 10.1016/j.edurev.2006.11.001.
- [101] V. Grubov, A. Badarin, and V. Maksimenko, "Analysis of information perception and processing during long-term and intense cognitive load using combined EEG and NIRS," in *Proc. Int. Conf. Nonlinearity, Inf. Robot. (NIR)*, Dec. 2020, pp. 12–13, doi: 10.1109/NIR50484.2020.9290234.
- [102] B. J. Morris, D. C. Willcox, T. A. Donlon, and B. J. Willcox, "NIH public access," *Gerontology*, vol. 61, no. 6, pp. 515–525, 2015.
- [103] N. Goukasian, S. Porat, A. Blanken, D. Avila, D. Zlatev, S. Hurtz, K. S. Hwang, J. Pierce, S. H. Joshi, E. Woo, and L. G. Apostolova, "Cognitive correlates of hippocampal atrophy and ventricular enlargement in adults with or without mild cognitive impairment," *Dementia Geriatric Cognit. Disorders Extra*, vol. 9, no. 2, pp. 281–293, Aug. 2019, doi: 10.1159/000490044.
- [104] C. Iadecola, "The pathobiology of vascular dementia," *Neuron*, vol. 80, no. 4, pp. 844–866, 2013, doi: 10.1016/j.neuron.2013.10.008.
- [105] A. J. Edwards, "Dementia and the patient-I," *Dementia*, pp. 137–165, Apr. 1993, doi: 10.1007/978-1-4757-9963-7_7.
- [106] K. P. Patel, D. T. Wymer, V. K. Bhatia, R. Duara, and C. D. Rajadhyaksha, "Multimodality imaging of dementia: Clinical importance and role of integrated anatomic and molecular imaging," *RadioGraphics*, vol. 40, no. 1, pp. 200–222, Jan. 2020, doi: 10.1148/rg.2020190070.
- [107] J. D. Oldan, V. L. Jewells, B. Pieper, and T. Z. Wong, "Complete evaluation of dementia: PET and MRI correlation and diagnosis for the neuroradiologist," *Amer. J. Neuroradiol.*, vol. 42, no. 6, pp. 998–1007, Jun. 2021, doi: 10.3174/ajnr.A7079.
- [108] Z. Arvanitakis, R. C. Shah, and D. A. Bennett, "Diagnosis and management of dementia: Review," *JAMA J. Amer. Med. Assoc.*, vol. 322, no. 16, pp. 1589–1599, 2019, doi: 10.1001/jama.2019.4782.
- [109] E. Razzicchia, P. Lu, W. Guo, and P. Kosmas, "A new metasurfaceenhanced microstrip patch antenna for haemorrhagic stroke detection," in *Proc. 15th Eur. Conf. Antennas Propag. (EuCAP)*, Mar. 2021, pp. 1–4, doi: 10.23919/EuCAP51087.2021.9410899.
- [110] C. A. Taylor, S. F. Greenlund, L. C. McGuire, H. Lu, and J. B. Croft, "Deaths from Alzheimer's disease—United States, 1999–2014," *Morbidity Mortality Weekly Rep.*, vol. 66, no. 20, pp. 521–526, May 2017, doi: 10.15585/mmwr.mm6620a1.
- [111] L. Nazarko, "Dementia 2. Alzheimer's disease: Diagnosis, treatment and managment," *Brit. J. Healthcare Assistants*, vol. 13, no. 7, pp. 329–335, Jul. 2019, doi: 10.12968/bjha.2019.13.7.329.
- [112] A. Porsteinsson, R. Isaacson, S. Knox, M. Sabbagh, and I. Rubino, "Diagnosis of early Alzheimer's disease: Clinical practice in 2021," *J. Prevention Alzheimer's Disease*, vol. 8, pp. 371–386, Jun. 2021, doi: 10.14283/jpad.2021.23.
- [113] I. Saied, M. S. R. Bashri, T. Arslan, C. Smith, and S. Chandran, "Dielectric measurements of brain tissues with Alzheimer's disease pathology in the microwave region," in *IEEE Proc. Int. Symp. Med. Meas. Appl. (MeMeA)*, Jun. 2019, pp. 1–6, doi: 10.1109/MeMeA.2019. 8802179.
- [114] M. S. R. Bashri and T. Arslan, "Low-cost and compact RF switching system for wearable microwave head imaging with performance verification on artificial head phantom," *IET Microw., Antennas Propag.*, vol. 12, no. 5, pp. 706–711, Apr. 2018, doi: 10.1049/iet-map.2017.0486.
- [115] M. S. R. Bashri, T. Arslan, and W. Zhou, "Flexible antenna array for wearable head imaging system," in *Proc. 11th Eur. Conf. Antennas Propag. (EUCAP)*, Mar. 2017, pp. 172–176, doi: 10.23919/EuCAP.2017.7928757.

- [116] A. Alqadami, A. Zamani, A. Trakic, and A. Abbosh, "Flexible electromagnetic cap for three-dimensional electromagnetic head imaging," *IEEE Trans. Biomed. Eng.*, vol. 68, no. 9, pp. 2880–2891, Sep. 2021, doi: 10.1109/TBME.2021.3084313.
- [117] A. Smida, "Simulation and analysis of variable antenna designs for effective stroke detection," *Int. J. Adv. Comput. Sci. Appl.*, vol. 11, no. 12, pp. 238–244, 2020, doi: 10.14569/IJACSA.2020.0111230.
- [118] J. A. T. Vasquez, R. Scapaticci, G. Turvani, G. Bellizzi, D. O. Rodriguez-Duarte, N. Joachimowicz, B. Duchêne, E. Tedeschi, M. R. Casu, L. Crocco, and F. Vipiana, "A prototype microwave system for 3D brain stroke imaging," *Sensors*, vol. 20, no. 9, pp. 1–16, 2020, doi: 10.3390/s20092607.
- [119] B. Sohani, B. Khalesi, N. Ghavami, M. Ghavami, S. Dudley, A. Rahmani, and G. Tiberi, "Detection of haemorrhagic stroke in simulation and realistic 3-D human head phantom using microwave imaging," *Biomed. Signal Process. Control*, vol. 61, Aug. 2020, Art. no. 102001, doi: 10.1016/j.bspc.2020.102001.
- [120] A. Patel, V. Berdunov, Z. Quayyum, D. King, M. Knapp, and R. Wittenberg, "Estimated societal costs of stroke in the U.K. based on a discrete event simulation," *Age Ageing*, vol. 49, no. 2, pp. 270–276, Feb. 2020, doi: 10.1093/ageing/afz162.
- [121] D. M. Oleske, X. Cheng, A. Jeong, and T. J. Arndt, "Pediatric acute ischemic stroke by age-group: A systematic review and meta-analysis of published studies and hospitalization records," *Neuroepidemiology*, vol. 55, no. 5, pp. 331–341, 2021, doi: 10.1159/000518281.
- [122] M. Kelly-Hayes, "Influence of age and health behaviors on stroke risk: Lessons from longitudinal studies," *J. Amer. Geriatrics Soc.*, vol. 58, no. 2, pp. 1–8, 2011, doi: 10.1111/j.1532-5415.2010.02915.x.
- [123] B. C. V. Campbell, D. A. De Silva, M. R. Macleod, S. B. Coutts, L. H. Schwamm, S. M. Davis, and G. A. Donnan, "Ischaemic stroke," *Nature Rev. Disease Primers*, vol. 5, no. 1, Dec. 2019, doi: 10.1038/s41572-019-0118-8.
- [124] J. Oliveira-Filho and W. J. Koroshetz, "Hemorrhagic stroke," in *Surgical Intensive Care Medicine*, 2nd ed. Berlin, Germany: Springer, 2010, pp. 163–171, doi: 10.1007/978-0-387-77893-8_16.
- [125] Strokes. Nidirect Government Services U.K. Accessed: Mar. 2022. [Online]. Available: https://www.nidirect.gov.uk/conditions/strokes
- [126] M. P. Lin and D. S. Liebeskind, "Imaging of ischemic stroke," *Continuum, Lifelong Learn. Neurol.*, vol. 22, no. 5, pp. 1399–1423, Oct. 2016, doi: 10.1212/CON.00000000000376.
- [127] B. J. Mohammed, A. M. Abbosh, S. Mustafa, and D. Ireland, "Microwave system for head imaging," *IEEE Trans. Instrum. Meas.*, vol. 63, no. 1, pp. 117–123, Jan. 2014, doi: 10.1109/TIM.2013.2277562.
- [128] A. S. M. Alqadami, A. E. Stancombe, K. S. Bialkowski, and A. Abbosh, "Flexible meander-line antenna array for wearable electromagnetic head imaging," *IEEE Trans. Antennas Propag.* vol. 69, no. 7, pp. 4206–4211, 2021.
- [129] B. Borja, J. A. Tirado, and H. Jardón, "An overview of UWB antennas for microwave imaging systems for cancer detection purposes," *Prog. Electromagn. Res. B*, vol. 80, pp. 173–198, 2018, doi: 10.2528/PIERB18030302.
- [130] H. A. Damis, R. Mirzavand, H. J. Chung, and P. Mousavi, "Flexible printed square loop antennas for wearable applications," in *Proc. 17th Int. Symp. Antenna Technol. Appl. Electromagn. ANTEM*, Jul. 2016, pp. 15–16, doi: 10.1109/ANTEM.2016.7550212.
- [131] W. Gao, Y. Zhu, Y. Wang, G. Yuan, and J.-M. Liu, "A review of flexible perovskite oxide ferroelectric films and their application," *J. Materiomics*, vol. 6, no. 1, pp. 1–16, Mar. 2020, doi: 10.1016/j.jmat.2019.11.001.
- [132] P. Sethi and S. R. Sarangi, "Internet of Things: Architectures, protocols, and applications," J. Electr. Comput. Eng., vol. 2017, pp. 1–25, 2017, doi: 10.1155/2017/9324035.
- [133] S. Huang, Y. Liu, Y. Zhao, Z. Ren, and C. F. Guo, "Flexible electronics: Stretchable electrodes and their future," *Adv. Funct. Mater.*, vol. 29, no. 6, Feb. 2019, Art. no. 1805924, doi: 10.1002/adfm. 201805924.
- [134] S. F. Jilani, Q. H. Abbasi, and A. Alomainy, "Inkjet-printed millimetrewave PET-based flexible antenna for 5G wireless applications," in *IEEE MTT-S Int. Microw. Symp. Dig.*, vol. 1, Aug. 2018, pp. 37–39, doi: 10.1109/IMWS-5G.2018.8484603.
- [135] Y. Liu, H. Wang, W. Zhao, M. Zhang, H. Qin, and Y. Xie, "Flexible, stretchable sensors for wearable health monitoring: Sensing mechanisms, materials, fabrication strategies and features," *Sensors*, vol. 18, no. 2, p. 645, Feb. 2018, doi: 10.3390/s18020645.

- [136] S. G. Kirtania, A. W. Elger, M. R. Hasan, A. Wisniewska, K. Sekhar, T. Karacolak, and P. K. Sekhar, "Flexible antennas: A review," *Micro-machines*, vol. 11, no. 9, pp. 1–43, 2020.
- [137] F. Albarracin-Vargas, F. Vega, C. Kasmi, F. AlYafei, and C. Baer, "Dual graded index dielectric lens system for hyperthermia," in *Proc. 33rd Gen. Assem. Scientific Symp. Int. Union Radio Sci.*, Aug. 2020, pp. 2020–2023, doi: 10.23919/URSIGASS49373.2020. 9232223.
- [138] D. Gharode, A. Nella, and M. Rajagopal, "State-of-art design aspects of wearable, mobile, and flexible antennas for modern communication wireless systems," *Int. J. Commun. Syst.*, vol. 34, no. 15, pp. 1–48, Oct. 2021, doi: 10.1002/dac.4934.
- [139] I. A. Tunio, Y. Mahe, T. Razban-Haghighi, and B. Froppier, "Mutual coupling reduction in patch antenna array using combination of shorting pins and metallic walls," *Prog. Electromagn. Res. C*, vol. 107, pp. 157–171, 2021, doi: 10.2528/PIERC20082803.
- [140] T. Subha, T. Subash, K. S. Claudia Jane, D. Devadharshini, and D. I. Francis, "Study and analysis of suppress of surface wave propagation in microstrip patch antenna," *Mater. Today, Proc.*, vol. 24, pp. 2414–2423, Jan. 2020, doi: 10.1016/j.matpr.2020.03.771.
- [141] R. Saleem, T. Shabbir, A. Quddus, F. Shafique, and U. Anwar, "Diversity/MIMO antenna incorporating electromagnetic band gap structures for isolation," in *Proc. Int. Conf. High Performance Comput. Simul. (HPCS)*, Jul. 2018, pp. 38–42, doi: 10.1109/HPCS.2018. 00020.
- [142] M. K. Khandelwal, B. K. Kanaujia, and S. Kumar, "Defected ground structure: Fundamentals, analysis, and applications in modern wireless trends," *Int. J. Antennas Propag.*, vol. 2017, pp. 1–22, 2017, doi: 10.1155/2017/2018527.
- [143] I. Nadeem and D.-Y. Choi, "Study on mutual coupling reduction technique for MIMO antennas," *IEEE Access*, vol. 7, pp. 563–586, 2019, doi: 10.1109/ACCESS.2018.2885558.
- [144] A. K. Arya, M. V. Kartikeyan, and A. Patnaik, "Defected ground structure in the perspective of microstrip antennas: A review," *Frequenz*, vol. 64, nos. 5–6, pp. 79–84, Jan. 2010, doi: 10.1515/FREQ.2010.64.5-6.79.
- [145] P. R. Prajapati, "Application of defected ground structure to suppress outof-band harmonics for WLAN microstrip antenna," *Int. J. Microw. Sci. Technol.*, vol. 2015, pp. 1–9, Dec. 2015, doi: 10.1155/2015/210608.
- [146] R. Anitha, V. P. Sarin, P. Mohanan, and K. Vasudevan, "Enhanced isolation with defected ground structure in MIMO antenna," *Electron. Lett.*, vol. 50, no. 24, pp. 1784–1786, 2014, doi: 10.1049/el.2014.2795.
- [147] B. A. Zeb, N. Nikolic, and K. P. Esselle, "A high-gain dualband EBG resonator antenna with circular polarization," *IEEE Antenna Wireless Propag. Lett.*, vol. 14, pp. 108–111, 2015, doi: 10.1109/LAWP.2014.2356599.
- [148] M. S. Alam, N. Misran, B. Yatim, and M. T. Islam, "Development of electromagnetic band gap structures in the perspective of microstrip antenna design," *Int. J. Antennas Propag.*, vol. 2013, pp. 1–22, Mar. 2013, doi: 10.1155/2013/507158.
- [149] D. M. Elsheakh, H. A. Elsadek, and E. A. Abdallah, "Antenna designs with electromagnetic band gap structures," in *Metamaterial*. London, U.K.: IntechOpen, 2012. [Online]. Available: https://www.intechopen.com/chapters/37016 doi: 10.5772/37222
- [150] C. Milias, R. B. Andersen, P. I. Lazaridis, Z. D. Zaharis, B. Muhammad, J. T. B. Kristensen, A. Mihovska, and D. D. S. Hermansen, "Metamaterial-inspired antennas: A review of the state of the art and future design challenges," *IEEE Access*, vol. 9, pp. 89846–89865, 2021, doi: 10.1109/ACCESS.2021.3091479.
- [151] M. Isabirye, D. V. Raju, M. Kitutu, V. Yemeline, J. Deckers, and J. Poesen, We are IntechOpen, the world's leading publisher of open access books built by scientists, for scientists TOP 1 %," Intech, London, U.K., Tech. Rep., 2012, p. 13, vol. 13, doi: 10.1039/C7RA00172J.
- [152] M. S. Khan, A. Capobianco, A. I. Najam, I. Shoaib, E. Autizi, and M. F. Shafique, "Compact ultra-wideband diversity antenna with a floating parasitic digitated decoupling structure," *IET Microw., Antennas Propag.*, vol. 8, no. 10, pp. 747–753, Jul. 2014, doi: 10.1049/iet-map.2013.0672.
- [153] K.-L. Wong, C.-Y. Tsai, and J.-Y. Lu, "Two asymmetrically mirrored gap-coupled loop antennas as a compact building block for eight-antenna MIMO array in the future smartphone," *IEEE Trans. Antennas Propag.*, vol. 65, no. 4, pp. 1765–1778, Apr. 2017.
- [154] L. Zhao, L. K. Yeung, and K. Wu, "A novel second-order decoupling network for two-element compact antenna arrays," in *Proc. Asia Pacific Microw. Conf.*, vol. 2, Dec. 2012, pp. 1172–1174.

- [155] L. K. Yeung and Y. E. Wang, "Mode-based beamforming arrays for miniaturized platforms," *IEEE Trans. Microw. Theory Techn.*, vol. 57, no. 1, pp. 45–52, Jan. 2009.
- [156] J. Andersen and H. Rasmussen, "Decoupling and descattering networks for antennas," *IEEE Trans. Antennas Propag.*, vol. 24, no. 6, pp. 841–846, Nov. 1976. 1976, pp. 841–846.
- [157] J. C. Coetzee, "Dual-frequency decoupling networks for compact antenna arrays," Int. J. Microw. Sci. Technol., vol. 2011, pp. 1–4, Dec. 2011, doi: 10.1155/2011/249647.
- [158] A. M. M. Dahlan and M. R. Kamarudin, "Shorted microstrip patch antenna with parasitic element," *J. Electromagn. Waves Appl.*, vol. 24, nos. 2–3, pp. 327–339, Jan. 2010, doi: 10.1163/ 156939310790735624.
- [159] K. D. Xu, J. Zhu, S. Liao, and Q. Xue, "Wideband patch antenna using multiple parasitic patches and its array application with mutual coupling reduction," *IEEE Access*, vol. 6, pp. 42497–42506, 2018, doi: 10.1109/ACCESS.2018.2860594.
- [160] S.-W. Su, C.-T. Lee, and F.-S. Chang, "Printed MIMO-antenna system using neutralization-line technique for wireless USB-dongle applications," *IEEE Trans. Antennas Propag.*, vol. 60, no. 2, pp. 456–463, Feb. 2012.
- [161] S. Wang and Z. Du, "Decoupled dual-antenna system using crossed neutralization lines for LTE/WWAN smartphone applications," *IEEE Antennas Wireless Propag. Lett.*, vol. 14, pp. 523–526, 2015, doi: 10.1109/LAWP.2014.2371020.
- [162] W. A. E. Ali and A. A. Ibrahim, "A compact double-sided MIMO antenna with an improved isolation for UWB applications," *AEU Int. J. Electron. Commun.*, vol. 82, pp. 7–13, Dec. 2017, doi: 10.1016/j.aeue.2017.07.031.
- [163] A. Petosa, Dielectric Resonator Antenna Handbook. Boston, MA, USA: Artech House, 2007.
- [164] C. A. Balanis, Antenna Theory: Analysis and Design, 3rd ed. Hoboken, NJ, USA: Wiley, 2005.
- [165] S. K. K. Dash, T. Khan, B. K. Kanaujia, and N. Nasimuddin, "Wideband cylindrical dielectric resonator antenna operating in HEM 11," *Int. J. Antennas Propag.*, vol. 2017, pp. 1–12, Jul. 2017.
- [166] H. A. Malhat, S. H. Zainud-Deen, H. El-Hemaily, H. A. Hamed, and A. A. Ibrahim, "Reconfigurable circularly polarized hemispherical DRA using plasmonic graphene strips for MIMO communications," *Plasmonics*, vol. 17, no. 2, pp. 765–774, Apr. 2022, doi: 10.1007/s11468-021-01581-9.
- [167] P. Anoop and R. Bhattacharjee, "Investigation on dual-band equilateral triangular shaped dielectric resonator antennas for WLAN applications," *Int. J. RF Microw. Comput.-Aided Eng.*, vol. 31, no. 7, pp. 1–15, Jul. 2021, doi: 10.1002/mmce.22672.
- [168] A. Mahmoud and H. Attia, "Wide-band circularly polarized dielectric resonator antenna array," in *Proc. IEEE Int. Symp. Antennas Propag.* USNC/URSI Nat. Radio Sci. Meeting, Jul. 2017, pp. 1521–1522, doi: 10.1109/APUSNCURSINRSM.2017.8072803.
- [169] Z. Chen, C. Shen, H. Liu, X. Ye, L. Qi, Y. Yao, J. Yu, and X. Chen, "Millimeter-wave rectangular dielectric resonator antenna array with enlarged DRA dimensions, wideband capability, and high-gain performance," *IEEE Trans. Antennas Propag.*, vol. 68, no. 4, pp. 3271–3276, Apr. 2020, doi: 10.1109/TAP.2019.2950101.
- [170] D. E. Anagnostou, M. T. Chryssomallis, B. D. Braaten, J. L. Ebel, and N. Sepúlveda, "Reconfigurable UWB antenna with RF-MEMS for ondemand WLAN rejection," *IEEE Trans. Antennas Propag.*, vol. 62, no. 2, pp. 602–608, Nov. 2014.
 [171] A. S. Kholapure and R. G. Karandikar, "Emerging techniques for printed
- [171] A. S. Kholapure and R. G. Karandikar, "Emerging techniques for printed reconfigurable antenna: A review," in *Proc. 2nd Int. Conf. Res. Comput. Intell. Commun. Netw. (ICRCICN)*, Sep. 2016, pp. 57–61.
- [172] R. J. Beneck, A. Das, G. Mackertich-Sengerdy, R. J. Chaky, Y. Wu, S. Soltani, and D. Werner, "Reconfigurable antennas: A review of recent progress and future prospects for next generation," *Prog. Electromagn. Res.*, vol. 171, pp. 89–121, 2021.
 [173] H. Joodaki, H. Valiee, and M. Bayat, "Reconfigurable dual fre-
- [173] H. Joodaki, H. Valiee, and M. Bayat, "Reconfigurable dual frequency microstrip MIMO patch antenna using RF MEMS switches for WLAN application," in *Proc. 25th Chin. Control Decis. Conf. (CCDC)*, May 2013, pp. 1–5.
- [174] S. Soltani, P. Lotfi, and R. D. Murch, "A port and frequency reconfigurable MIMO slot antenna for WLAN applications," *IEEE Trans. Antennas Propag.*, vol. 64, no. 4, pp. 1209–1217, Apr. 2016.
- [175] E. Vassos, J. Churm, J. Powell, C. Viegas, B. Alderman, and A. Feresidis, "Air-bridged Schottky diodes for dynamically tunable millimeter-wave metamaterial phase shifters," *Sci. Rep.*, vol. 11, no. 1, pp. 1–10, Dec. 2021, doi: 10.1038/s41598-021-85565-z.

- [176] M. Pastorino, *Microwave Imaging*. Hoboken, NJ, USA: Wiley, May 2010, pp. 251–256, doi: 10.1002/9780470602492.
- [177] S. Dumanli, "Challenges of wearable antenna design," in *Proc.* 46th Eur. Microw. Conf. (EuMC), Oct. 2016, pp. 1350–1352, doi: 10.1109/EuMC.2016.7824602.
- [178] C. Hu, H. Zuo, and Y. Li, "Effects of radiofrequency electromagnetic radiation on neurotransmitters in the brain," *Frontiers Public Health*, vol. 9, pp. 1–15, Aug. 2021, doi: 10.3389/fpubh.2021.691880.
- [179] W.-J. Zhi, L.-F. Wang, and X.-J. Hu, "Recent advances in the effects of microwave radiation on brains," *Mil. Med. Res.*, vol. 4, no. 1, pp. 1–14, Dec. 2017, doi: 10.1186/s40779-017-0139-0.
- [180] J. Schüz, G. Waldemar, J. H. Olsen, and C. Johansen, "Risks for central nervous system diseases among mobile phone subscribers: A Danish retrospective cohort study," *PLoS One*, vol. 4, no. 2, pp. 1–5, 2009, doi: 10.1371/journal.pone.0004389.
- [181] O. P. Gandhi, "Microwave emissions from cell phones exceed safety limits in Europe and the U.S. when touching the body," *IEEE Access*, vol. 7, pp. 47050–47052, 2019, doi: 10.1109/ACCESS.2019.2906017.
- [182] EPSRC COG-MHEAR Research Program. Accessed: Apr. 2022. [Online]. Available: http://cogmhear.org
- [183] N. Gillani and T. Arslan, "Intelligent sensing technologies for the diagnosis, monitoring and therapy of Alzheimer's disease: A systematic review," *Sensors*, vol. 21, no. 12, p. 4249, Jun. 2021, doi: 10.3390/s21124249.
- [184] N. N. Kucherov, M. A. Deryabin, and M. G. Babenko, "Homomorphic encryption methods review," in *Proc. IEEE Conf. Russian Young Researchers Electr. Electron. Eng. (EIConRus)*, Jan. 2020, pp. 370–373, doi: 10.1109/EIConRus49466.2020.9039110.
- [185] G. Peralta, R. G. Cid-Fuentes, J. Bilbao, and P. M. Crespo, "Homomorphic encryption and network coding in IoT architectures: Advantages and future challenges," *Electron.*, vol. 8, no. 8, pp. 1–14, 2019, doi: 10.3390/electronics8080827.
- [186] M. J. Sheller, B. Edwards, G. A. Reina, J. Martin, S. Pati, A. Kotrotsou, M. Milchenko, W. Xu, D. Marcus, R. R. Colen, and S. Bakas, "Federated learning in medicine: Facilitating multi-institutional collaborations without sharing patient data," *Sci. Rep.*, vol. 10, no. 1, pp. 1–12, Dec. 2020, doi: 10.1038/s41598-020-69250-1.
- [187] N. Rieke, J. Hancox, W. Li, F. Milletarì, H. R. Roth, S. Albarqouni, S. Bakas, M. N. Galtier, B. A. Landman, K. Maier-Hein, S. Ourselin, M. Sheller, R. M. Summers, A. Trask, D. Xu, M. Baust, and M. J. Cardoso, "The future of digital health with federated learning," *npj Digit. Med.*, vol. 3, no. 1, pp. 1–7, Dec. 2020, doi: 10.1038/s41746-020-00323-1.
- [188] T. Justinia and K. Saud, "Blockchain technologies: Opportunities for solving real-world problems in healthcare and biomedical sciences," *Acta Informat. Medica*, vol. 27, no. 9, pp. 284–291, 2022, doi: 10.5455/aim.2019.27.284-291.
- [189] S. P. Mann, J. Savulescu, P. Ravaud, and M. Benchoufi, "Blockchain, consent and prosent for medical research," *J. Med. Ethics*, vol. 47, no. 4, pp. 244–250, May 2020, doi: 10.1136/medethics-2019-105963.
- [190] T. Cover and P. Hart, "Nearest neighbor pattern classification," *IEEE Trans. Inf. Theory*, vol. 13, no. 1, pp. 21–27, Jan. 1967, doi: 10.1109/TIT.1967.1053964.
- [191] M. Elsaadouny, J. Barowski, and I. Rolfes, "Humanitarian microwave imaging enhancement and classification of shallowly buried objects," in *Proc. IEEE 10th Annu. Inf. Technol., Electron. Mobile Commun. Conf. (IEMCON)*, Oct. 2019, pp. 394–397, doi: 10.1109/IEMCON.2019.8936165.
- [192] T. Hacib, Y. Le Bihan, M. R. Mekideche, H. Acikgoz, O. Meyer, and L. Pichon, "Microwave characterization using least-square support vector machines," *IEEE Trans. Magn.*, vol. 46, no. 8, pp. 2811–2814, Aug. 2010, doi: 10.1109/TMAG.2010.2043657.
- [193] S. P. Rana, M. Dey, G. Tiberi, L. Sani, A. Vispa, G. Raspa, M. Duranti, M. Ghavami, and S. Dudley, "Machine learning approaches for automated lesion detection in microwave breast imaging clinical data," *Sci. Rep.*, vol. 9, no. 1, pp. 1–12, Dec. 2019, doi: 10.1038/s41598-019-46974-3.
- [194] B. Gerazov and R. C. Conceic, "Deep learning for tumour classification in homogeneous breast tissue in medical microwave imaging," in *Proc. IEEE EUROCON 17th Int. Conf. Smart Technol.*, Ohrid, Macedonia, Jul. 2017, pp. 6–8
- [195] H. M. E. Misilmani and T. Naous, "Machine learning in antenna design: An overview on machine learning concept and algorithms," in *Proc. Int. Conf. High Perform. Comput. Simul.* (*HPCS*), Jul. 2019, pp. 600–607, doi: 10.1109/HPCS48598.2019. 9188224.

- [196] L. Harrison, M. Ravan, K. Zhang, and R. K. Amineh, "Identification of materials using a microwave sensor array and machine learning," in *Proc. Int. Appl. Comput. Electromagn. Soc. Symp. (ACES)*, 2021, 2021. [Online]. Available: https://ieeexplore.ieee.org/document/9528672
- [197] M. M. Khan, S. Hossain, P. Mozumdar, S. Akter, and R. H. Ashique, "A review on machine learning and deep learning for various antenna design applications," *Heliyon*, vol. 8, no. 4, Apr. 2022, Art. no. e09317, doi: 10.1016/j.heliyon.2022.e09317.
- [198] B. C.-K. Tong, "HHS public access," *Physiol. Behav.*, vol. 176, no. 5, pp. 139–148, 2017, doi: 10.1109/TAP.2020.2978952.
- [199] R. Ullah and T. Arslan, "PySpark-based optimization of microwave image reconstruction algorithm for head imaging big data on highperformance computing and Google cloud platform," *Appl. Sci.*, vol. 10, no. 10, p. 3382, May 2020, doi: 10.3390/APP10103382.
- [200] R. Ullah and T. Arslan, "Parallel delay multiply and sum algorithm for microwave medical imaging using spark big data framework," *Algorithms*, vol. 14, no. 5, p. 157, May 2021, doi: 10.3390/a14050157.
 [201] P. Liu, C. Rapaport, Y. Z. Wei, and S. Sridhar, "Simulated biological
- [201] P. Liu, C. Rapaport, Y. Z. Wei, and S. Sridhar, "Simulated biological materials at microwave frequencies for the study of electromagnetic Hyperthermia," in *IEEE EMBS Digest*, 1992, pp. 272–273.
- [202] S. Costanzo, V. Cioffi, A. M. Qureshi, and A. Borgia, "Gel-like human minicking phantoms: Realization procedure, dielectric characterization and experimental validations on microwave wearable body sensors," *Biosensors*, vol. 11, no. 4, p. 111, Apr. 2021, doi: 10.3390/bios11040111.
- [203] B. Mohammed, K. Białkowski, S. Hill, A. Stancombe, A. Alqadami, M. T. Heitzmann, and A. Abbosh, "Stable and lifelong head phantoms using polymer composition mimicking materials to test electromagnetic medical imaging systems," *IEEE J. Electromagn., RF Microw. Med. Biol.*, vol. 5, no. 4, pp. 322–328, Dec. 2021, doi: 10.1109/JERM.2021.3051311.
- [204] N. Joachimowicz, B. Duchêne, C. Conessa, and O. Meyer, "Anthropomorphic breast and head phantoms for microwave imaging," *Diagnostics*, vol. 8, no. 4, p. 85, Dec. 2018, doi: 10.3390/diagnostics8040085.



USMAN ANWAR received the B.S. degree in electrical (telecommunication) engineering from the National University of Computer and Emerging Sciences (NUCES), FAST, Islamabad, Pakistan, in 2010, and the M.S. degree in telecommunication engineering from the University of Engineering and Technology, Taxila, Pakistan, in 2016. He is currently pursuing the Ph.D. degree with the School of Engineering, The University of Edinburgh, Edinburgh, U.K., with a research focus

on neurodegeneration detection through RF and microwave sensors. He has an experience of over ten years in the telecommunication industry, he worked as an Assistant Manager with Pakistan Telecommunication Company Ltd., from 2011 to 2016; and the Manager, from 2016 to 2021. He was responsible for managing the nationwide fiber optic backbone network in close liaison with Nokia, Huawei, ZTE, and FiberHome. He has worked on major optical fiber communication technologies such as DWDM, SDH, OTN, MPLS-TP, and PTN. His research interests include microwave and millimeter-wave antennas, MIMO, ultra-wideband MIMO, and biomedical sensors.



TUGHRUL ARSLAN holds the Chair of Integrated Electronic Systems with the School of Engineering, The University of Edinburgh, Edinburgh, U.K. He is a member of the Integrated Micro and Nano Systems (IMNS) Institute and leads the Embedded Mobile and Wireless Sensor Systems (EWireless) Group. He is the author of more than 500 refereed papers and the inventor of more than 20 patents. His research interests include developing low-power radio frequency sensors for

wearable and portable biomedical applications. He is a member of the IEEE CAS Executive Committee on VLSI Systems and Applications, in 1999; and

a member of the steering and technical committees of several international conferences. He is the Co-Founder of the NASA/ESA Conference on Adaptive Hardware and currently serves as a member for its steering committee. He is currently an Associate Editor of the IEEE TRANSACTIONS ON VERY LARGE SCALE INTEGRATION (VLSI) SYSTEMS and was previously an Associate Editor of the IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS—I: REGULAR PAPERS, from 2005 to 2006; and IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS—II: EXPRESS BRIEFS, from 2008 to 2009.



AMIR HUSSAIN (Senior Member, IEEE) received the B.Eng. (Hons.) and Ph.D. degrees from the University of Strathclyde, Glasgow, in 1992 and 1997, respectively. He is the Found-ing Director of the Centre of AI and Data Science, Edinburgh Napier University, U.K. He has (co)authored several patents and over 500 publications, including over 250 journal articles and over 20 books/monographs. He has led major national and international projects, including the

current (\$4 Million) EPSRC funded COG-MHEAR Grant and supervised over 35 Ph.D. students. His research interests include cross-disciplinary and industry-led, with a focus on cognitive data science and trustworthy AI technologies to engineer smart and secure healthcare and industrial systems of tomorrow. Among other distinguished roles, he is a member of the U.K. Computing Research Committee (the national expert panel of the IET and the BCS for U.K. computing research), the General Chair of IEEE WCCI 2020 (the world's largest IEEE technical event in computational intelligence, comprising IJCNN, IEEE CEC, and FUZZ-IEEE), the Chair of the IEEE U.K. and Ireland Chapter of the IEEE Industry Applications Society, the Vice-Chair of the Emergent Technologies Technical Committee of the IEEE Computational Intelligence Society (CIS), the IEEE U.K. and Ireland Chapter Chair of the Industry Applications Society, and the (Founding) Vice-Chair of the IEEE CIS Task Force on Intelligence Systems for e-Health. He is the Founding Chief Editor of the journal Cognitive Computation (Springer) (IF: 5.4) and the book series Socio-Affective Computing (Springer). He has served as an Invited Associate Editor/Editorial Board Member for various other journals, including the IEEE TRANSACTIONS ON NEURAL NETWORKS AND LEARNING SYSTEMS, Information Fusion (Elsevier), the IEEE TRANSACTIONS ON SYSTEMS, MAN, and CYBERNETICS: SYSTEMS, and the IEEE TRANSACTIONS ON EMERGING TOPICS IN COMPUTATIONAL INTELLIGENCE.



PETER LOMAX holds the position of Senior Experimental Officer with The University of Edinburgh, managing the Scottish Microelectronics Centre within the School of Integrated Micro and Nano Systems (IMNS). Involved principally in research projects involving micro and nanofabrication technologies. He has 24 years of industrial and academic experience, ranging from a radio operator on an Antarctic research base to the principal investigator (PI) on research projects. His research

interests include X-ray emitter fabrication and MEMS/nano detectors.